



Optimal climate policy as if the transition matters

Emanuele Campiglio, Simon Dietz and Frank Venmans

December 2022

Centre for Climate Change Economics and Policy Working Paper No. 412 ISSN 2515-5709 (Online)

Grantham Research Institute on Climate Change and the Environment Working Paper No. 387 ISSN 2515-5717 (Online)



Economic and Social Research Council

. The Grantham Foundation







The Centre for Climate Change Economics and Policy (CCCEP) was established by the University of Leeds and the London School of Economics and Political Science in 2008 to advance public and private action on climate change through innovative, rigorous research. The Centre is funded by the UK Economic and Social Research Council. Its third phase started in October 2018 with seven projects:

- 1. Low-carbon, climate-resilient cities
- 2. Sustainable infrastructure finance
- 3. Low-carbon industrial strategies in challenging contexts
- 4. Integrating climate and development policies for 'climate compatible development'
- 5. Competitiveness in the low-carbon economy
- 6. Incentives for behaviour change
- 7. Climate information for adaptation

More information about CCCEP is available at www.cccep.ac.uk

The Grantham Research Institute on Climate Change and the Environment was established by the London School of Economics and Political Science in 2008 to bring together international expertise on economics, finance, geography, the environment, international development and political economy to create a world-leading centre for policy-relevant research and training. The Institute is funded by the Grantham Foundation for the Protection of the Environment and a number of other sources. It has 13 broad research areas:

- 1. Biodiversity
- 2. Climate change adaptation and resilience
- 3. Climate change governance, legislation and litigation
- 4. Climate, health and environment
- 5. Environmental behaviour
- 6. Environmental economic theory
- 7. Environmental policy evaluation
- 8. International climate politics
- 9. Science and impacts of climate change
- 10. Sustainable public and private finance
- 11. Sustainable natural resources
- 12. Transition to zero emissions growth
- 13. UK national and local climate policies

More information about the Grantham Research Institute is available at www.lse.ac.uk/GranthamInstitute

Suggested citation:

Campiglio E, Dietz S and Venmans F (2022) Optimal climate policy as if the transition matters. Centre for Climate Change Economics and Policy Working Paper 413/Grantham Research Institute on Climate Change and the Environment Working Paper 387. London: London School of Economics and Political Science

This working paper is intended to stimulate discussion within the research community and among users of research, and its content may have been submitted for publication in academic journals. It has been reviewed by at least one internal referee before publication. The views expressed in this paper represent those of the authors and do not necessarily represent those of the host institutions or funders.

Optimal climate policy as if the transition matters^{*}

Emanuele Campiglio^{a,b,c}, Simon Dietz^{c,d}, and Frank Venmans^{c,e}

^a University of Bologna, Department of Economics, Bologna, Italy

^bRFF-CMCC European Institute on Economics and the Environment (EIEE), Milan, Italy

^cLondon School of Economics and Political Science, UK ^dCESifo, Munich, Germany

^e University of Mons, Belgium

This version: December 2022

Abstract

The optimal transition to a low-carbon economy must account for adjustment costs in switching from dirty to clean capital, technological progress, and economic and climatic shocks. We study the low-carbon transition using a dynamic stochastic general equilibrium model with emissions abatement costs calibrated on a large energy modelling database, solved with recursive methods. We show how capital inertia puts upward pressure on emissions and temperatures in the short run, but that nonetheless it is optimal to actively disinvest from – to 'strand' – a significant share of the dirty capital stock. Conversely, clean technological progress, as well as uncertainty about climatic and economic factors, lead to lower emissions and temperatures in the long run. Putting these factors together, we estimate a net premium of 33% on the optimal carbon price today relative to a 'straw man' model with perfect capital mobility, fixed abatement costs and no uncertainty.

^{*}Email for correspondence: F.Venmans1@lse.ac.uk. We would like to thank Ghassane Benmir, Yongyang Cai, Derek Lemoine, Rick van der Ploeg, Ivan Rudik and participants at the National Bank of Belgium 2020 conference, IAEE 2021, EAERE 2021, RES 2021, AERE 2021, EPSF 2022 as well as seminar participants at SWEEET, Paris School of Economics, and the Potsdam Institute for Climate Impact Research. This research was originally funded by the National Bank of Belgium. EC also gratefully acknowledges the financial support of the European Research Council (ERC) under the European Union's Horizon 2020 Research and Innovation Programme (Grant agreement No. 853050 - SMOOTH). SD and FV gratefully acknowledge funding from the ESRC Centre for Climate Change Economics and Policy (ref. ES/R009708/1), and the Grantham Foundation for the Protection of the Environment. The authors have no conflicts of interest or financial interests to disclose.

Keywords: adjustment costs; carbon price; climate change; low-carbon transition; stranded assets; technological progress; uncertainty *JEL codes*: C61, E22, H23, O44, Q54, Q55

1 Introduction

In the long run, greenhouse gas emissions will have to be virtually eliminated for global temperatures to stop rising (IPCC, 2021). Almost all countries have signed up to the long-run temperature goals of the 2015 UN Paris Agreement on climate change, namely "Holding the increase in the global average temperature to well below 2°C above pre-industrial levels and pursuing efforts to limit the temperature increase to 1.5°C" (UNFCCC, 2016).¹ Although these are fundamentally political goals, there is also broad agreement in economics research with the basic premise that emissions should fall from their current record levels to very low levels eventually (Howard and Sylvan, 2021).

However, how best to make the *transition* from a high- to low-carbon economy remains open for debate. How high should initial carbon prices/taxes be, and how should they change over time (Golosov et al., 2014; Lemoine and Rudik, 2017b; Gerlagh and Liski, 2018; Dietz and Venmans, 2019; Daniel et al., 2019; Mattauch et al., 2020)? Should we make immediate, deep emissions cuts, or should we postpone our efforts until later, in effect using up the remaining carbon budget more quickly (van der Ploeg, 2018; Gollier, 2021)? Can fossil-fuel-burning assets be used until the end of their economic lifetimes, or do we need to leave them 'stranded'² (McGlade and Ekins, 2015; Barnett, 2019; van der Ploeg and Rezai, 2020b; Cahen-Fourot et al., 2021)?

One reason why the debate remains open is that views differ about the optimal long-run temperature that would maximise social welfare (Hänsel et al., 2020). But another reason why these questions endure is that making the transition to a low-carbon economy involves complex dynamics that go beyond basic models of emissions abatement. In this paper, we develop a dynamic stochastic general equilibrium model with the aim of analysing these complex transition dynamics. The model exhibits three key features that, we contend, have not been studied in a single model before.

First, we capture limits to how fast emissions can be reduced through the introduction of two kinds of adjustment cost (Gould, 1968; Lucas Jr, 1967). The first kind are convex

¹The 2021 Glasgow Climate Pact increased the emphasis on the more ambitious 1.5°C goal (UNFCCC, 2021).

 $^{^{2}\}text{A}$ rapid transition could lead carbon-intensive capital to be decommissioned prematurely, used at a lower capacity rate, or retrofitted or repurposed (e.g., via carbon capture and storage technologies) at a cost, with a monetary loss for capital owners.

investment costs to capital accumulation (clean or dirty). These capture the additional costs faced when rapidly accumulating new capital and lead to smoother investment paths, as large capital investments concentrated in a short period are more expensive than the same capital expansion spread over a longer period. The second kind are convex costs of dirty capital disinvestment. To rapidly reduce emissions, it may be optimal to dispose of dirty capital faster than its natural depreciation rate. It is often assumed that capital is perfectly mobile (as are other production factors), so it can be costlessly shifted from dirty to clean sectors. But some capital has been designed to work in tandem with fossil fuels (e.g., coal and gas power plants, steel mills, ships, etc.) and would be costly to convert to low-carbon uses, perhaps prohibitively so. The costs of a rapid transition would be transmitted by production and financial networks, and it has even been suggested that a rapid transition could trigger economic and financial instability (Semieniuk et al., 2021). We assume *physical* asset stranding (i.e., disinvestment from dirty capital) generates convex *monetary* stranding costs: the more rapid is the disinvestment of dirty physical capital, the larger is the proportion of the monetary value of dirty capital that gets lost.

Second, we incorporate the widely observed phenomenon of falling clean technology costs. Like adjustment costs, these can affect the timing of investment in clean capital. Clean technological progress has diverse sources, including knowledge spillovers from other sectors such as artificial intelligence and nanotechnology, economies of scale and learning-by-doing in the clean technology sector, induced R&D into clean technology, and network effects (e.g., relating to charging electric vehicles). In our model, falling clean technology costs are partly autonomous/exogenous and partly induced, i.e., an endogenous function of cumulative emissions abatement, as implied by evidence on experience/learning curves (Boston Consulting Group, 1970; Weiss et al., 2010). Endogenous clean technological progress creates an incentive to abate earlier, unlike exogenous technological progress. In many basic models of emissions abatement, clean technology costs are either fixed or if they fall then the process is fully exogenous (Gillingham et al., 2008).

Third, charting an optimal transition path is fraught with uncertainties, including about economic growth prospects, technological progress, the physical climate system, and the impacts of climate change (see Heal and Millner, 2014; Cai, 2021, for reviews). The optimal transition should account for current uncertainties but also how they change over time. Policy-makers will figure things out along the way. We incorporate multiple sources of uncertainty, including the business cycle, climate variability, and rare macroeconomic shocks (Barro, 2009) made more likely by climate change. These uncertainties affect the optimal transition, which is formulated as a stochastic welfare maximisation problem and solved using recursive dynamic programming methods so that the planner anticipates and reacts to shocks.³ In doing so, we employ Epstein-Zin-Weil preferences (Epstein and Zin, 1989; Weil, 1990), which disentangle risk aversion from intertemporal consumption substitution, consistent with empirical data on asset returns. By contrast, most basic models compute deterministic optimal paths, or at most they are subjected to sensitivity analysis, computing multiple deterministic optimal paths, while varying the values of key exogenous parameters (Lemoine and Rudik, 2017a).

We obtain analytical solutions for optimal clean and dirty capital accumulation, as well as for the optimal carbon price, but our main findings are obtained by solving the model numerically. We calibrate emissions abatement costs by fitting the model on a large database of results from detailed energy models using the generalised method of moments. We find the optimal transition to a low-carbon economy is fast. From 2020 to 2030, emissions are roughly halved, whether we consider a pure economic planning problem of maximising the discounted net benefits of emissions reductions, or adopt the international climate policy approach of meeting an exogenous constraint on warming of 1.5°C at the lowest discounted abatement cost. Either way, a high carbon price is required, as is active 'stranding' of dirty assets, rather than working with the grain of capital depreciation. If the imposition of a carbon price is delayed and its eventual introduction is not anticipated, even more dirty assets are stranded.

Disentangling the effects of capital inertia, clean technological progress, and uncertainty, we show how adjustment costs prevent an implausibly large optimal shift from dirty to clean capital initially, and how capital inertia therefore results in temperature inertia and higher carbon prices. Falling clean technology costs lead to lower optimal emissions and temperatures in the long run, while also allowing for modestly lower optimal carbon prices all along the path. The price effect is the net effect of exogenous technological progress, which allows lower prices, and endogenous technological progress, which requires a carbon price premium to the extent that innovation externalities are not internalised using other policy instruments. Uncertainty leads to lower optimal emissions and temperature-related macro-economic disaster risk. Putting these three effects together, we estimate a net premium of 33% on the optimal carbon price today relative to a 'straw man' model with perfect capital mobility, fixed abatement costs and no uncertainty. The premium falls to about 20% by mid-century. Higher prices pay off in terms of lower temperatures, but only in the long run, specifically the second half of this century, because of capital inertia.

Lastly, rather than comparing an unconstrained cost-benefit problem with a temperature-

 $^{^{3}}$ In other words, optimal emissions are set by a closed-loop or feedback policy, where emissions are a function of future states of the coupled climate-economy system (Lemoine and Rudik, 2017a).

constrained cost-effectiveness problem, as above, we compare two different ways to stay below the exogenous temperature constraint: (i) maximising discounted net benefits of emissions reductions, and (ii) minimising discounted abatement costs. The latter stays closer to how the problem is defined in international climate policy. We make the point that ignoring climate damages leads the planner to put off – to back-load – emissions reductions. We estimate the optimal carbon price is 13% lower initially. Since climate damages are not zero, we conclude the latter approach underestimates optimal carbon prices.

Our paper contributes to the literature studying optimal global climate policies using Integrated Assessment Models or IAMs, which was pioneered by William Nordhaus (Nordhaus, 1991; Nordhaus and Boyer, 2000; Nordhaus, 2008, 2013). The focus on *optimal* policies contrasts with the large applied literatures on the 'social cost of carbon' (i.e., marginal damage cost of CO_2) along arbitrary emissions paths (Tol, 2011), and on the marginal abatement cost of meeting pre-determined climate goals (Clarke et al., 2014). While most of the IAM literature has computed deterministic optimal paths, the relevance of uncertainty has led to the recent development of a class of 'recursive IAMs', which compute an adaptive optimal path in the face of unfolding uncertainties using recursive methods (e.g. Lemoine and Traeger, 2014; Cai and Lontzek, 2019; van den Bremer and van der Ploeg, 2021; Karydas and Xepapadeas, 2022). Our model belongs to this class, which has mostly focused on quantifying the effects of sources of uncertainty (Lemoine and Rudik, 2017a, review the literature). We connect this work with a smaller and more recent literature looking at adjustment costs along the low-carbon transition (Coulomb et al., 2019; Hambel et al., 2020; van der Ploeg and Rezai, 2020a; Vogt-Schilb et al., 2018). This line of work is motivated in part by an interest in whether pricing carbon will leave dirty capital assets stranded (also see Baldwin et al., 2020; Rozenberg et al., 2020), but is mostly deterministic.

The remainder of the paper is structured as follows. Section 2 presents the model and section 3 briefly presents some of the key mechanics of the optimal transition. Section 4 explains our calibration strategy. Section 5 presents and discusses our numerical results. Section 6 offers a discussion of several key issues, including further comparisons with the literature, and Section 7 concludes.

2 The model

For expositional reasons, consider an economy in continuous time with an infinite horizon. We suppress time subscripts unless time-dependence is not obvious. A single type of good Y exists, used for both consumption and investment. The production function is

$$Y = AL^{1-\alpha}K^{\alpha}\Lambda\Omega,\tag{1}$$

where A is total factor productivity, L is labour, K is capital, and α is the capital share. Labour increases exogenously at constant rate g_L . The multiplier Λ represents the costs of emissions abatement. Similarly, Ω is the climate damage multiplier.

Abatement costs

The abatement cost multiplier is

$$\Lambda = \exp\left(-\frac{\varphi_t}{2}\mu^2\right),\tag{2}$$

where μ stands for abatement, which implies a marginal abatement cost (MAC) function

$$-Y_{\mu}/Y = \varphi_t \mu, \tag{3}$$

with slope φ_t . Note the slope parameter is time-dependent. Abatement is the difference between business-as-usual (BAU) emissions, which specifically means the level of emissions at a carbon price of zero, and actual emissions:

$$\mu = \phi_t / \varphi_t - E. \tag{4}$$

The parameter ϕ_t represents the MAC at zero emissions and thereby contributes to determining the level of BAU emissions ϕ_t/φ_t . Bounding the zero-emissions MAC at ϕ_t is consistent with evidence on and assumptions about zero-carbon technology costs, e.g., direct air capture of CO₂ (Heutel et al., 2016). By contrast, some model structures imply an infinite MAC at zero emissions, e.g., the commonly used model that differentiates between clean and dirty capital/energy inputs and assumes a constant elasticity of substitution between them.

Clean technological progress

In the model, clean technology costs fall over time. Part of this process is exogenous and part of it depends endogenously on emissions abatement. Exogenous cost decreases could result from innovation spillovers elsewhere in the economy (Popp, 2019; Taalbi, 2020). For example, improvements in lithium-ion batteries in the smartphone industry are now benefiting electric mobility, thin-layer nanotechnology might be useful for photo-voltaic energy on plastics, and chemical membrane technology could be employed to produce "green hydrogen" from water. They could also result from basic R&D directly into clean technology that is independent of levels of clean technology use. Endogenous cost decreases could come from learning-bydoing, network effects, or other sources of increasing returns, as well as induced R&D that is proportional to the size of the clean technology market (Acemoglu et al., 2012).

The MAC at zero emissions evolves according to

$$\phi_t = \phi_0 \left(\omega \frac{1}{M^{\epsilon}} + (1 - \omega) \frac{1}{1 + g_{\phi} \tau} \right), \tag{5}$$

and similarly the MAC function slope parameter evolves according to

$$\varphi_t = \varphi_0 \left(\omega \frac{1}{M^{\epsilon}} + (1 - \omega) \frac{1}{1 + g_{\varphi} \tau} \right).$$
(6)

Thus, the overall decrease in clean technology costs is a weighted average of the exogenous and endogenous processes, where ω is the relative weight on endogenous cost reductions.

To represent exogenous cost decreases, the parameters of the MAC function are a decreasing function of time. The exogenous evolution of the intercept term ϕ_t is given by $\phi_0/(1 + g_{\phi}\tau)$ and the slope term by $\varphi_0/(1 + g_{\varphi}\tau)$. τ is just a monotonic transformation of time, $\tau = 1 - \exp(-g_{\tau}t)$, $g_{\tau} > 0$, which is a technical detail necessary to solve the model numerically (Traeger, 2014). One can think of it as artificial time, starting at zero and gradually approaching one in the long term. The important point is that ϕ_t starts at ϕ_0 and decreases towards a long run asymptote $\phi_0/(1 + g_{\phi})$. The same goes for φ_t . We assume $g_{\phi} = g_{\varphi}$, which implies constant BAU emissions given (4), consistent with the latest evidence on future emissions conditional on current policies (IPCC, 2022), where current policies still involve trivially low pricing of global aggregate emissions (World Bank, 2021).

For endogenous technological progress, M represents cumulative emissions abatement relative to time zero⁴ and ϵ is the constant elasticity of the MAC function parameters with respect to cumulative emissions. This means that for every doubling of cumulative emissions, MACs are reduced by the same factor $1-2^{-\epsilon}$, i.e., we assume a constant rate of cost reduction.

Damages and warming

The damage multiplier

$$\Omega = \exp\left(-\frac{\gamma}{2}T^2\right),\tag{7}$$

⁴Cumulative abatement in the model is $M_0 + \int_0^t \mu_s ds$, where M_0 is cumulative abatement at time zero. Hence, we can define cumulative abatement relative to time zero as $M(S,t) = [M_0 + E_{BAU}t - (S_t - S_0)]/M_0$, where S stands for cumulative emissions.

where γ is the damage function coefficient and T is global mean temperature relative to its pre-industrial level.

Climate science has shown that T is approximately linearly proportional to cumulative carbon dioxide emissions (IPCC, 2021), so we write

$$T = \zeta S,\tag{8}$$

where S stands for cumulative greenhouse gas emissions and ζ is the slope coefficient, known as the transient climate response to cumulative carbon emissions or TCRE. Although temperature responds to a CO₂ emission impulse with a delay of a few years, the relationship between temperature and cumulative emissions is still well approximated by Eq. (8) (Dietz and Venmans, 2019). Thus, as Dietz and Venmans (2019) and Dietz et al. (2021a) showed, not only is (8) simple, it mimics the behaviour of complex climate science models much better than most climate modules in IAMs that explicitly model the atmospheric stock of carbon and thereby require more state variables.

Temperature is subject to a geometric Brownian motion capturing uncertainty in the temperature impulse response to emissions,

$$dT = \zeta E dt + \sigma_T T dW_T, \tag{9}$$

where emissions times the TCRE parameter ζ is by definition the drift and the temperature volatility is σ_T . The Brownian motion represents both epistemic uncertainty, i.e., the fact that our climate model is imperfect, and aleatoric uncertainty, i.e., genuinely random volatility of the climate system.⁵.

Clean and dirty capital

The capital stock is composed of dirty and clean technologies. Dirty capital K_d represents all forms of capital requiring fossil-fuel inputs (e.g., coal- and gas-fired power plants, integrated steel mills, and cement plants).⁶ Clean capital K_c represents capital that does not directly require fossil fuels for production. Upstream sectors where fossil fuels are a direct input

$$dT = \zeta E dt + \frac{\sigma_1}{\zeta} T dW_1 + \sigma_2 T dW_2, \tag{10}$$

which is identical to our equation with $\sigma_T = \frac{\sigma_1}{\zeta} + \sigma_2$.

⁵This could be explicitly modelled as follows: assume epistemic uncertainty (Bayesian updating) of the TCRE follows the process $d\zeta = \sigma_1 dW_1$ and aleatoric (weather) shocks are proportional to warming, adding an uncorrelated geometric Brownian motion with volatility σ_2 . This results in

⁶Or which otherwise emit greenhouse gases as a by-product, as cement and some chemical plants do, for example.

need to shift towards low-carbon production technologies. Downstream sectors where fossil fuels are not directly used can already be considered clean. They are dirty only insofar as their intermediate inputs come from fossil-fuel-burning upstream sectors, e.g., electricity (Kemp-Benedict, 2018). Therefore, rather than being a small expanding niche in a system dominated by dirty capital, our definition implies most of the capital stock is clean (or at least not dirty *per se*), and we treat as dirty only the capital that directly uses fossil fuels or otherwise directly emits greenhouse gases.

Use of K_d produces emissions E at intensity ψ_t ,

$$E = \psi_t K_d. \tag{11}$$

The emissions intensity of dirty capital ψ_t decreases at rate g_{ψ} , which represents autonomous improvements in energy efficiency that occur without carbon pricing due to incentives to save on energy costs. This is distinct from falling clean technology costs.⁷

Use of K_c does not produce any emissions. Substituting dirty capital with clean capital reduces emissions at a cost to output (and reduces climate damages). The technical rate of substitution is

$$\frac{Y_{K_d}}{Y_{K_c}} = 1 + \frac{(\phi_t \psi_t - \varphi_t \psi_t^2 K_d)(K_d + K_c)}{\alpha},$$
(12)

which is decreasing in the stock of dirty capital assuming $K_d < K_c$, meaning that the less dirty capital there is, the more clean capital is required at the margin to keep output constant. At BAU emissions, $K_d = \phi_t/(\varphi_t \psi_t)$ and both forms of capital are equally productive,⁸ whereas at zero emissions dirty capital is several times more productive than clean capital (determined by ϕ_t). Substitutability is further impaired by adjustment costs, to which we now turn.

Capital accumulation, adjustment and disinvestment costs

Clean and dirty capital are expanded by investment, but this is subject to convex increasing adjustment costs:

$$\iota_j = \frac{\chi_j i_j^2}{k}, \ j \in \{c, d\},$$
(13)

where we use lower case to denote units of effective labour (see Appendix A), thus *i* is gross investment per unit of effective labour in clean and dirty capital respectively, $k = k_c + k_d$

⁷More specifically, we define $\psi_t = \psi_0 \frac{1+\tau}{e^{(g+g_L)t}}$.

⁸Although our model has constant BAU emissions, BAU dirty capital is initially *increasing at rate* g_{ψ} due to the effect of decreasing emissions intensity. By contrast, BAU dirty capital as a proportion of GDP equals $\frac{K_{d,BAU}}{Y} = \frac{\phi_0}{Y_0\varphi_0\psi_0(1+\tau)}$ and is *decreasing* to half its initial value in the long run. With constant emissions intensity, this proportion of dirty capital would decrease at rate $g + g_L$ and become zero in the steady state.

is total capital per unit of effective labour, and χ_j is the adjustment cost parameter. For any amount i_j of disbursed investment expenditure, the value of newly installed capital will only be a proportion of investment expenditure equal to $i_j - (\chi_j i_j^2)/k$. Adjustment costs are proportional to total capital such that the dynamics of total capital are homogeneous of degree one.⁹

In addition to adjustment costs of positive investment in clean/dirty capital, we also model the costs of negative investment or disinvestment of dirty capital, which may be optimal in order to make a rapid transition, rather than waiting for dirty capital to depreciate.¹⁰ Denoting negative dirty investment $r \equiv i_d$ iff $i_d < 0$,

$$\kappa(r) = \frac{\theta_1 r^{\theta_2}}{k},\tag{14}$$

where the cost function has coefficient $\theta_1 > 0$ and exponent $\theta_2 > 1$. When dirty assets are disinvested and retrofitted/repurposed, strictly less than 100% of the value of those assets can be recovered. Thus, those dirty assets are devalued and become partly 'stranded'. For low r, a large share of the value of the original dirty capital can be recovered. But as r increases, the share of the dirty capital assets' value that is lost increases more than proportionally.

Both capital stocks are subject to a geometric Brownian motion with correlation coefficient ρ_k . This stochastic process represents random changes to the value of capital and its productivity, similar to changes in the valuation of firms on the stock market, and allows us to capture part of the uncertainty about future growth prospects.

Putting these elements together, the equations of motion of dirty and clean capital are

$$dk_j = \left(i_j - \frac{\chi_j i_j^2}{k} - (\delta_j + g_L + g)k_j\right) dt + \sigma_j k_j dW_j,$$
(15)

where δ_j is the depreciation rate and $dW_c = \rho_k dW_d + (1 - \rho_k^2)^{1/2} dW_2$ with $dW_2 \perp dW_d$.

At zero dirty investment, emissions naturally decrease at rate $\delta_d + g_{\psi}$. This can be interpreted as the 'speed limit' to decarbonisation, beyond which dirty capital needs to be repurposed at a cost.¹¹ This penalty on rapid abatement is important in determining the cost of an abrupt transition to a low-carbon economy.

⁹Note that one cannot divide by k_d because the ratio would converge to zero.

¹⁰Given we start in a high-carbon economy, the opposite case $i_c < 0$ is of limited relevance.

¹¹Analytically, we can see this by expressing the (initial) speed of abatement as $\frac{\dot{E}}{E} = g + g_L - g_{\psi} + \frac{\dot{k_d}}{k_d} = -\delta_d - g_{\psi} + \frac{i_d - \chi_d i_d^2/k}{k_d}$. The exact formula for all future periods is more convoluted but with the same interpretation.

Disasters and tipping points

TFP follows a geometric Brownian motion capturing normal economic fluctuations. In addition, TFP is subject to rare macroeconomic disasters, such as wars, financial crises and pandemics (Barro, 2009). But, in a twist on the original formulation of these shocks, we specify them as an increasing function of temperature to account for the risk of climaterelated shocks, e.g., due to crossing tipping points in the physical climate system (Cai and Lontzek, 2019; Dietz et al., 2021b). TFP therefore evolves according to

$$d\tilde{A} = \left(g_{A1}\tilde{A} + g_{A2}(\tilde{A}_0 - \tilde{A})\right)dt + \sigma_A\tilde{A}dW_A - \xi_1(1 + \xi_2 T)\tilde{A}dP,$$
(16)

where \tilde{A} is TFP adjusted for production per unit of effective labour,¹² The trend growth rate g_{A1} is set so that $g_{A1}\tilde{A}$ corresponds to the expected effect of the shocks (as in Barro, 2009), while g_{A2} governs a very slow return to a balanced growth path. σ_A is 'normal' economic volatility, ξ_1 is the size of the macro-economic shock in the absence of climate change (from Barro, 2009) and ξ_2 parameterises how this increases as a function of temperature. dP is a Poisson process and equals one with probability λ_A .

Consumption and utility

Agents have rational expectations. Consumption per unit of effective labour is

$$c = y - i_d - i_c - \kappa(r). \tag{17}$$

Disinvestment costs reduce the income available from liquidating dirty capital (or, equivalently, retrofitting/repurposing costs must be paid out of income). In order to disentangle risk aversion from intertemporal consumption smoothing, we use the recursive preference formulation of Duffie and Epstein (1992), which is equivalent to the discrete-time recursive preferences of Epstein and Zin (1989), Kreps and Porteus (1978), and Weil (1990). The representative household maximises utility from per-capita consumption, discounted at the pure time preference rate ρ , with standardised aggregator function f and value function V:

$$V(k_d, k_c, S, \tilde{A}, t) = \max_{i_c, i_d} \mathbb{E} \int_t^\infty -f(k_d, k_c, S, \tilde{A}, t, V, i_d, i_c) ds,$$
(18)

with

$$f(k_d, k_c, S, \tilde{A}, t, V, i_d, i_c) = \frac{c^{1-\eta}}{1-\eta} \Upsilon - \hat{\rho} V \frac{1-RRA}{1-\eta},$$
(19)

 ${}^{12}\tilde{A} = Ae^{(\alpha-1)gt}$ and production per unit of effective labour $y = \tilde{A}k^{\alpha}\Lambda\Omega$.

$$\Upsilon = \hat{\rho} \left(\left(1 - RRA \right) V \right)^{\frac{\eta - RRA}{1 - RRA}}, \qquad (20)$$

and

$$\hat{\rho} = \rho - g_L + (\eta - 1)g.$$
(21)

 η is the elasticity of marginal utility of consumption (equal to the inverse of the elasticity of intertemporal substitution) and *RRA* is the coefficient of relative risk aversion. Setting $\eta = RRA$ would simplify the expression leading back to the standard expected utility case. However, this would be difficult to reconcile with empirical data on risky and risk-free asset returns, and in the context of climate policy it has the property that high levels of risk aversion also increase the discount rate and lead to lower climate ambition, all else being equal.

3 Optimal capital accumulation and carbon pricing

In this short section, we build intuition for our quantitative results below by looking at how adjustment costs affect the optimal accumulation of clean and dirty capital, and how adjustment costs, clean technological progress and uncertainty affect the optimal pricing/taxation of greenhouse gas emissions.

The Hamilton-Jacobi-Bellman (HJB) equation is

$$\max_{i_c, i_d} \left\{ f + \frac{1}{\mathrm{d}t} \mathbb{E}[dV(k_d, k_c, S, \tilde{A}, t)] \right\} = 0.$$
(22)

Applying Itô's Lemma (see Appendix A) and maximising the HJB equation with respect to the capital stocks yields expressions for the shadow prices of clean and dirty capital:

$$V_{k_c} = c^{-\eta} \Theta \frac{k}{k - 2\chi_c i_c}$$
⁽²³⁾

and

$$V_{k_d} = c^{-\eta} \Theta\left(\frac{k}{k - 2\chi_d i_d}\right) \left(1 - \frac{\theta_1 \theta_2 r^{(\theta_2 - 1)}}{k}\right),\tag{24}$$

where $c^{-\eta}$ is the marginal utility of consumption, adjusted for Duffie/Epstein preferences via

$$\Theta = \hat{\rho} \left(\left(1 - RRA \right) V \right)^{1 - \frac{1 - \eta}{1 - RRA}}.$$
(25)

Combining (23) and (24) into a ratio of the shadow prices of clean and dirty capital helps

make the role of adjustment costs clear:

$$\frac{V_{k_c}}{V_{k_d}} = \frac{k - 2\chi_d i_d}{k - 2\chi_c i_c} \left(\frac{1}{1 - (\theta_1 \theta_2 r^{(\theta_2 - 1)})/k} \right)$$
(26)

In the absence of adjustment costs, the ratio of the shadow prices of clean and dirty capital would be one, as the opportunity cost of investing an extra dollar is the utility of consuming that dollar instead, regardless of where the investment goes. However, adjustment costs change this. The first element on the right-hand side is the ratio of adjustment costs of dirty and clean capital investment. The ratio is increasing in clean capital investment and decreasing in dirty capital investment. Moreover, if we assume $\chi_c = \chi_d$, as we do below, then the ratio just depends on the absolute amount of investment going into each form of capital. Thus, adjustment costs put a brake on how quickly capital accumulation can shift from dirty to clean. The second element of the ratio is the costs of disinvesting dirty capital. The shadow price of clean capital is relatively larger, the larger these dirty capital disinvestment costs are. For high levels of disinvestment, i.e., $r > \left(\frac{\theta_1\theta_2}{k}\right)^{\frac{1}{1-\theta_2}}$, this factor becomes negative: dirty capital becomes a burden, because the costs of disinvestment exceed its initial value.

Applying the envelope theorem to the HJB equation, we can also obtain an analytic expression for the optimal marginal abatement cost. For simplicity, we call this the optimal 'carbon price', but note that if endogenous clean technological progress is affected by market imperfections other than climate damages themselves, the first best would be to internalise these using complementary policy instruments such as innovation subsidies. The optimal carbon price is:

$$p = \Phi \{ \Psi c^{-\eta} y \gamma S - V_{SS} E$$
 marginal dama,

$$+ \Psi c^{-\eta} y \left[\frac{\varphi}{2} \mu^2 (1 - \omega) \frac{\epsilon}{M^{\epsilon+1}} \right] - V_{tS}$$
 technological cl

$$- V_{Sk_d} \left[i_d - \chi_d i_d^2 - (\delta_d + g_L + g) k_d \right]$$
 dirty capital co

$$- V_{Sk_c} \left[i_c - \chi_c i_c^2 - (\delta_c + g_L + g) k_c \right]$$
 clean capital ex

$$- V_{SS} S \sigma_S^2 - \frac{1}{2} V_{SSS} S^2 \sigma_S^2$$
 temperature ris

$$- \frac{1}{2} V_{Sk_d k_d} k_d^2 \sigma_d^2 - \frac{1}{2} V_{Sk_c k_c} k_c^2 \sigma_c^2 - V_{Sk_c k_d} k_d k_c \rho^k \sigma_c \sigma_d$$
 capital risk pre

$$- \frac{1}{2} V_{SAA} \tilde{A}^2 \sigma_A^2$$
 TFP volatility

$$- \lambda_A \left[V_S (\tilde{A} - \Delta \tilde{A}) - V_S (\tilde{A}) \right]$$
 disaster risk

$$+ \lambda_A V_A (\tilde{A} - \Delta \tilde{A}) \xi_1 \xi_2 \zeta \tilde{A} - V_{SA} \left(\tilde{A} g_{A1} + \left(A_0 - \tilde{A} \right) g_{A2} \right) \},$$
 premium

marginal damage cost sechnological change effect dirty capital contraction effect clean capital expansion effect semperature risk premium capital risk premiums FFP volatility risk premium disaster risk premium (27)

where

$$\Phi = \frac{(1-\eta) L_0 \exp\left[(g+g_L) t\right]}{\hat{\rho} \left[1 - RRA + (RRA - \eta) c^{1-\eta} \left((1 - RRA) V\right)^{-\frac{1-\eta}{1-RRA}}\right] V_{k_c}}$$
(28)

and

$$\Psi(V) = \frac{\hat{\rho}}{\left(\left(1 - RRA\right)V\right)^{\frac{1 - \eta}{1 - RRA} - 1}}.$$
(29)

To interpret Eq. (27), bear in mind that $-V_S$ is the optimal carbon price in utils and is positive. This helps sign the numerous cross-derivatives.

The term on the first row of (27) is the marginal damage cost of an emission, comprising the current level of marginal damages $\Psi c^{-\eta} y \gamma S$ and the future change in marginal damages $-V_{SS}E = -V_{SS}\dot{T}/\zeta$. On a path of rising temperatures, a convex damage function ensures rising marginal damages, i.e., $-V_{SS} > 0$, which is true of our damage function (7) like most others.¹³

The term on the second row is the technical change effect. The first element is the marginal gain of endogenous technical change, which increases the optimal carbon price because the prospect of reducing the MAC creates an added incentive to abate emissions today. Building on our observation above, part of this incentive may be external to innovating firms. If another policy instrument exists to internalise this externality, then less than the full marginal gain of endogenous technical change should be factored into the *explicit* carbon price levied on the economy. However, there are many drivers of endogenous technology cost reductions (learning-by-doing, network effects, induced R&D, etc.), so in reality the distinction is likely to be less than clear-cut. The second element, V_{tS} , embodies the two exogenous, time-dependent processes that reduce the MAC in the model, i.e., falling clean technology costs and autonomous improvements in energy efficiency. These reduce the optimal carbon price by making emissions abatement exogenously cheaper. Note that V_{tS} is also affected by endogenous technological change, $\partial M/\partial t = E_{BAU} > 0$, i.e., observing the same level of cumulative emissions in a later period implies higher cumulative abatement. This will also reduce the optimal carbon price.

The terms on the third and fourth rows capture, respectively, the effect of shrinking dirty capital on the optimal carbon price, and the effect of expanding clean capital. In both cases, the effect of a marginal emission on the opportunity cost of capital (V_{Sk_d} and V_{Sk_c}) is positive, because the marginal emission increases temperatures along the path, higher temperatures reduce consumption, and lower consumption results in higher marginal utility,

¹³Technological change uncertainty will affect the impact of rising damages V_{SS} . Therefore, we will compare the total effect on the carbon price in the results section

thus a higher opportunity cost. Dirty capital is in decline along the optimal path, so the trend effect $[i_d - \chi_d i_d^2 - (\delta_d + g_L + g)k_d]$ is negative and overall the dirty capital contraction effect reduces the optimal carbon price. Decreasing dirty capital reduces future warming. Clean capital is expanding along the optimal path, with a positive trend effect and thus the optimal carbon price is reduced.

The remaining rows include the risk premiums corresponding to the four stochastic processes in the model: uncertainty about the temperature impulse response to emissions (the TCRE), uncertainty about capital valuations, uncertainty about TFP due to normal economic fluctuations/volatility, and the effect of uncertain disasters. These all increase the optimal carbon price.¹⁴ The disaster risk premium has two components. The first is the difference a disaster makes to marginal damages. On the one hand, marginal damages are smaller in a smaller, post-disaster economy. On the other hand, marginal utility is higher in a smaller economy. The latter effect outweighs the former when $\eta > 1$. The second component of the disaster risk premium takes account of the fact that disasters are larger at higher temperatures.

In Section 5, we quantitatively decompose our optimal carbon price trajectory into these different components.

4 Calibration

We calibrate our model using one-year time steps, with g_{τ} set equal to 0.015^{15} . The initial period of our numerical simulations is 2020. Population is measured in billion people; capital stock and output in trillion 2020US\$; GHG emissions in gigatonnes of CO2 equivalent (GtCO2eq); and temperature in degrees Celsius (°C).

We adopt three main strategies to calibrate our parameters. A first subset of parameters is defined by adopting standard values in the literature or recent empirical estimates. A second subset is calibrated to replicate empirical evidence or desired features of starting values. Finally, a third subset of particularly important parameters is estimated by fitting our model to a large database of results from detailed energy systems models using maximum likelihood methods. Table A1 in Appendix B provides an overview of the numerical values used for the parameters and initial conditions.

¹⁴Notice that since temperature is linearly proportional to cumulative emissions, the Brownian motion on temperature can be rewritten as a Brownian motion on S instead.

¹⁵This value was chosen so that $g_{\tau} = g + g_L - g_{\psi}$, which facilitates the convergence of the model algorithm.

4.1 Preferences

We use the expert survey of Drupp et al. (2018) to set both the elasticity of marginal utility of consumption (i.e., the inverse of the elasticity of intertemporal substitution) $\eta = 1.35$ and the utility discount rate $\rho = 1.1\%$. The coefficient of relative risk aversion *RRA* is set equal to 4, following Barro (2009).

4.2 Production

We calibrate the starting value of TFP $A_0 = 3.44$ so that, given initial values for labour and the capital stock, the 2020 value of global GDP is US\$84.5 trillion (World Bank, 2022). The TFP growth rate compensating disasters is set to $g_{A1} = 0.0037$, with fluctuations following a Brownian motion with standard deviation $\sigma_A = 0.013$ (van den Bremer and van der Ploeg, 2021), and the steady state stabilisation rate (which has a negligible effect) $g_{A2} = 0.01$. These values allow us to obtain a mean economic growth rate of 2% in the steady state. We follow UN (2019) in setting the initial value of population $L_0 = 7.795$ and population growth $g_L = 0.42\%$.¹⁶ The share of capital in production is assumed to be 0.3, as is common in the literature. We set the initial value of the aggregate capital stock, $K_0 = \text{US}$ \$348 trillion, close to its steady state value,¹⁷ determined by the parameter values and equal to a fixed proportion of output.¹⁸

To identify the shares of clean and dirty capital, we use the World Input-Output Database (WIOD), containing sectoral capital stock data for 43 countries at a NACE level 2 disaggregation (Timmer et al., 2015). As discussed in Section 2, we interpret dirty capital as the upstream productive assets requiring fossil fuels to operate. We thus label certain WIOD sectors as dirty because of their fossil-dependent technological basis (e.g., mining, coke, chemicals, plastics, metals, electricity, transportation). Where possible, we employ data from the Exiobase multi-regional input-output database (Stadler et al., 2018) to further disaggregate stocks and isolate those directly related to fossil fuels (e.g., the capital stock used for the exploration and extraction of coal, gas and oil) from those with less of a connection to fossil fuels (e.g., mining of non-ferrous metals and other materials). To perform this disaggregation, we assume that the share of fossil-related capital stock of a specific sector is equal to the share of fossil-related output in total sectoral output. With this method, we estimate that approximately 8% of the capital stock is dirty. We thus set $K_{d,0} = US$28 trillion and$

¹⁷The steady state level of total capital is given by $K^* = L \left(A e^{-\frac{\gamma}{2}T^{*2}} \frac{\alpha}{\rho + \delta + \eta g} \right)^{\frac{1}{1-\alpha}}$.

¹⁶This is equal to the average growth rate in the UN population medium variant scenario for the 2020-2100 period.

¹⁸In other words, we assume that the steady state capital-output ratio, equal to 3.7 in our central parameter simulations, also applies to the neighbourhood of the steady state.

 $K_{c,0} = \text{US}$ \$320 trillion.

The depreciation rate of dirty capital δ_d plays a particularly important role in our model as, together with the decline rate of emissions intensity ψ , it defines the maximum speed at which the system can abate emissions without having to disinvest from dirty capital. We assume δ_d is equal to the inverse of the average asset lifetime. Since K_d captures a wide variety of productive stocks with different asset lifetimes,¹⁹ we choose a central value of 4%, corresponding to an average asset lifetime of 25 years. This is roughly in line with depreciation rates adopted by models including long-lived physical assets.²⁰ For simplicity and to facilitate the dynamics of the model, we also set $\delta_c = \delta_d$. Given our initial value of $K_d = \text{US}$ \$28 trillion, and 2020 GHG emissions of 56 GtCO2eq, we set the initial value of emissions intensity ψ_0 equal to 2 GtCO2eq/trillion US\$. We assume emissions intensity declines at a rate approximately equal to its historical decline rate $g_{\psi} = 0.01$ (World Bank, 2022).²¹

A wide range of estimates exist for the adjustment cost parameter (see for instance Cooper and Haltiwanger, 2006; David and Venkateswaran, 2019). We follow van der Ploeg and Rezai (2020a) in setting a mid-range value of $\chi_d = \chi_c = 0.1$.

Both capital stocks are subject to Brownian motions, with standard deviations $\sigma_d = \sigma_c = 0.013$. The combined volatility of TFP and capital mimics the GDP volatility in van den Bremer and van der Ploeg (2021). In addition to capital volatility, our model also includes the possibility of macroeconomic disasters, which we assume to lead to a loss of 20% of GDP with a 1.7% probability, thus setting $\xi_1 = 0.2$. This is taken from Barro (2009), with a downward adjustment to reflect the fact that Barro estimated the size of regional disasters whereas our model is globally aggregated.

4.3 Climate and emissions

IPCC (2021) estimates the temperature increase in 2020 relative to pre-industrial $T_0 = 1.2^{\circ}$ C,

¹⁹Coal plants, for instance, have an average lifetime of 46 years (Cui et al., 2019), with many of them lasting up to 60 or 70 years. Similarly, IEA (2019) uses a reference lifetime of coal plants of 50 years. Other carbon-related assets like transport infrastructure and buildings have even longer life-spans, reaching 100 or 200 years, while manufacturing equipment is estimated to last between 10 and 40 years (Hallegatte, 2009; IEA, 2002).

 $^{^{20}}$ For instance, van der Ploeg and Rezai (2020a) use a depreciation rate of 5% for fossil-fuel exploration capital; Coulomb et al. (2019) use 3.33% for electricity generation plants; Baldwin et al. (2020) use 5% for the general capital stock and 2.5% for fossil energy capital; Vogt-Schilb et al. (2018) use 4% for industry and 2.5% for energy capital.

 $^{^{21}}$ We calculate emissions intensity using total GHG emissions (net of agricultural methane and nitrous oxide emissions) and value added in the industry sector (including construction). We then calculate the average emissions intensity decline using data from 1994 to 2007, to avoid an excessive distortionary effect of climate policy on the data).

which we use. IPCC (2021) also offers a central estimate of the TCRE ζ of 0.00045°C/TtCO2. However, this estimate does not take into account the 'Zero Emissions Commitment' to warming,²² and excludes some known climate feedbacks. Taking these into account, and including the warming effect of non-CO2 emissions, we estimate $\zeta = 0.0006.^{23}$ The combination of T_0 and ζ also allows us to calibrate the starting value of cumulative emissions $S_0 = 2000$. The parameter γ governing the climate damage function is set equal to 0.0077, based on the meta-analysis of Howard and Sterner (2017).

Temperature is subject to a Brownian motion with a standard deviation $\sigma_T = 0.03$, calibrated to fit the range of simulated temperatures at the end of this century across the IPCC climate models.²⁴ Thus, it represents uncertainty about the mechanisms linking emissions to temperature change.

Finally, the parameter governing the dependence of macro disasters to temperature ξ_2 is set to 0.07. This assumes an increase of the macro disaster jump from 20% to 21.4% when T is equal to 2°C.

4.4 Abatement, disinvestment and technical change

We estimate a set of key parameters linked to abatement and disinvestment costs (φ_0 , θ_1 , g_{φ} and ϵ) using energy systems modelling results, in particular from: (i) the model/scenario ensemble underpinning the IPCC Special Report on Global Warming of 1.5°C (IPCC, 2018); and (ii) the model/scenario ensemble of the Network of Central Banks and Supervisors for Greening the Financial System or NGFS (Bertram et al., 2021).²⁵ These databases together span 18 leading energy systems models.

Each model or project has a BAU reference scenario, and these scenarios differ depending on assumptions about population growth, GDP growth, openness of economies and other variables. Various emissions abatement scenarios are included, with a resulting range of temperature increases from below 1.5°C to more than 3°C. After excluding scenarios with temperatures too high to be compatible with meaningful emissions abatement,²⁶ we are left with 134 scenarios, for which we have values for carbon prices, as well as for emissions and

 $^{^{22}}$ That is, the unrealised warming to which we are committed due to CO2 already emitted to the atmosphere.

 $^{^{23}}$ The cumulative CO2 emission budget for 0.73°C warming from 2020 onwards is 1000 GtCO2 (IPCC, 2021). We assume that 20% of warming stems from other greenhouse gases.

²⁴Specifically, in 2081-2100 for all combinations of Representative Concentration Pathways (RCPs) and Shared Socioeconomic Pathways (SSPs), as reported by IPCC (2021).

²⁵The databases are available at https://data.ene.iiasa.ac.at/iamc-1.5c-explorer and https://www.ngfs.net/ngfs-scenarios-portal/, respectively.

 $^{^{26}}$ We exclude scenarios with cumulative emissions of more than 2166GtCO2eq in 2100, corresponding to a temperature increase of above 2.5°C.

GDP, from which we can calculate the scenarios' total abatement costs. The calibration strategy is to simultaneously choose the values of φ_0 , θ_1 , g_{φ} and ϵ that maximise the fit of our model to a combination of the marginal and total abatement costs from the scenario database.

As mentioned in Section 2, BAU emissions are given by $E_{BAU} = \phi_t/\varphi_t$, and since ϕ_t and φ_t decrease at the same rate, BAU emissions are constant. We use this to set $\phi = \varphi E_{BAU}$, with $E_{BAU} = 60$ GtCO2eq.²⁷ From (1), production under BAU emissions is $Y_{BAU} = AL^{1-\alpha}K^{\alpha} \exp\left(-\frac{\gamma}{2}T^2\right)$, thus allowing us to define total abatement costs as a percentage of GDP as $1 - \frac{Y}{Y_{BAU}}$. In turn, we can write

$$\frac{Y}{Y_{BAU}} = \exp\left[-\frac{\varphi_t}{2}\left(E - \frac{\phi_t}{\varphi_t}\right)^2\right] - \theta_1 r^{\theta_2} \frac{L_0 e^{(g+g_L)t}}{k^* Y_{BAU}},\tag{30}$$

where $r = \frac{-\dot{E} + E\left(-\delta - g - g_L + g_\tau \frac{1-\tau}{1+\tau}\right)}{\psi_0 L_0(1+\tau)}$ is the optimal disinvestment from dirty capital and k^* is the steady state total capital value from the model. Using the MAC function (3), marginal abatement costs are

$$-Y_{\mu} = Y\left(\phi_t - \varphi_t E\right) \tag{31}$$

where the MAC, $-Y_{\mu}$, is the optimal carbon price.

Parameters φ_0 , θ_1 , g_{φ} and ϵ are estimated by the Generalised Method of Moments using (30) and (31) simultaneously. Fitting to both total and marginal abatement costs allows us to obtain more robust results, in case the MAC functions of the underlying models are nonlinear.²⁸ We truncate the sample to eliminate outliers, excluding the 5% highest and lowest costs in each period. We put double weight on model results from the NGFS database. This database includes many of the models in the IPCC database, but since it is more recent it incorporates improvements in those models, e.g., calibration on the latest evidence.

Table 1 reports the results. We assume the errors of both equations are normally distributed. We report estimation results assuming technology costs that decrease exogenously, and technology costs that decrease endogenously. We do not have precise available estimates for ω , governing the relative weight of exogenous and endogenous technical change. Taalbi (2020) finds that at least 30% of variations in innovation activity are driven by the stimulus

²⁷This is 4GtCO2eq above the 2020 level of GHG emissions of 56GtCO2eq, which itself was negatively impacted by the Covid-19 pandemic, perhaps by up to 2GtCO2eq (IPCC, 2022). Thus, we assume policies actually implemented to price carbon, implicitly or explicitly, reduced GHG emissions by at least 2GtCO2eq below BAU in 2020.

 $^{^{28}}$ The MAC is the most economically meaningful cost measure, because it determines the first-order conditions of the optimum. Therefore, we use a diagonal weighting matrix with a 60% weight on the marginal abatement cost fit and 40% weight on the total abatement cost fit.

Table 1: GMM estimates of total abatement costs (equation 30) and marginal abatement costs (equation 31) simultaneously. The database consists of 122 scenarios of the IPCC Special Report on 1.5°C (excluding scenarios with cumulative emissions beyond 2500GtCO2) and 12 scenarios from the NGFS database. BIC: Bayesian Information Criterion. AIC: Akaike Information Criterion.

| | Exogenous technology | Endogenous technology | | |
|-----------------------------|----------------------|-----------------------|--|--|
| φ_0 | 4.2e-05*** | 4.2e-05*** | | |
| $	heta_1$ | $7.0e + 02^{**}$ | 7.1e+02** | | |
| g_arphi | 0.74^{***} | | | |
| ϵ | | 0.10*** | | |
| | | | | |
| Ν | 1605 | 1605 | | |
| Log likelihood | 6664 | 6660 | | |
| BIC | -13305 | -13297 | | |
| AIC | -13321 | -13313 | | |
| * p<.1; ** p<.05; *** p<.01 | | | | |

coming from the network of other productive activities. We opt for a mid-value of 0.5.

 g_{ϕ} is by design set to be equal to g_{φ} . The 2020 cumulative abatement value M_0 is set at an approximate value of 100 GtCO2, building on the information provided by UNEP (2021).²⁹ Finally, θ_2 is set to 2.1 to ensure our convex function has a continuous second derivative.

5 Results

In this section, we present our main results. We show the optimal transition to a low-carbon economy in different versions of the model, focusing on four key variables: (i) GHG emissions (E); (ii) global mean temperatures (T); (iii) carbon prices (p); and (iv) disinvestment from dirty capital assets (R). For (i) and (ii), we take a century-long perspective, whereas for the economic variables (iii) and (iv) we only look out as far as 2050, so that important near-term differences are easier to see.

We consider two different policy/planning objectives. The first is the economist's natural objective of maximising welfare as per Eq. (18), i.e., maximising the discounted net benefits of reducing GHG emissions through shifting from dirty to clean capital, trading off avoided climate damages with the costs of reducing emissions. We call this the 'cost-benefit' problem for short. The second objective places a constraint on expected warming of 1.5°C and minimises only discounted emissions abatement costs, rather than minimising the sum of

 $^{^{29}}$ We slightly increase the values from UNEP (2021) to improve the log likelihood fit of the model.

discounted emissions abatement costs and climate damages.³⁰ We call this the '1.5°C costeffectiveness' problem for short and include it because it is more true to the international climate negotiations in reality.

Technical details of how we solve the model can be found in Appendix C.

5.1 The optimal transition

Figure 1 shows the optimal transition to a low-carbon economy in the full model, with the cost-benefit solution in red and the 1.5°C cost-effectiveness solution in blue. The cost-benefit solution involves immediate, steep GHG emissions reductions. Optimal emissions are halved in each of the 2020s and the 2030s. Thereafter, the rate of emissions decrease slows. As a consequence of the steep initial drop in emissions, expected warming is limited to 1.65°C above pre-industrial in 2050 and 1.76°C in 2100, which is approximately peak warming.

To deliver steep optimal emissions reductions, the planner levies a high initial carbon price of US\$133/tCO2eq, rising to US\$283/tCO2eq in 2050. Pricing carbon at this level triggers an initial US\$375 billion/year in disinvestment from dirty capital assets, decreasing to US\$213 billion/year in 2030 and US\$80bn/year in 2040. Cumulatively, US\$4.8 trillion of disinvestment from dirty capital is made, which equates to 17% of the dirty capital stock at the start. Thus, the pace of the optimal transition requires not only an immediate halt to dirty capital investment but also active decommissioning or repurposing of dirty assets, rather than using them until the end of their erstwhile economic lifetimes. In Section 2, we characterised an emissions reduction speed limit of $\delta_d + g_{\psi}$, equal to 5%/year according to our calibration. This is how fast emissions would decrease if investment in dirty capital were zero, there were no dirty capital disinvestment, and reductions were thus left to a combination of capital depreciation and autonomous improvements in energy efficiency. The optimal transition involves breaking this speed limit, reducing emissions by 6.5-7%/year for the next two decades. This additional disinvestment creates cumulative stranding costs of US trillion, i.e., roughly 1/3 of the monetary value of dirty capital that is disinvested is lost.

Despite rapidly reducing emissions, the welfare-maximising path in Figure 1 does not limit warming to 1.5°C above pre-industrial. Therefore, keeping warming to no more than 1.5°C requires even steeper emissions reductions, not so much in the 2020s but increasingly thereafter. Indeed, emissions are slightly negative for most of the second half of the

³⁰Technically, this is achieved by maximising (18) with a damage function (7) that is almost discontinuous, i.e., a ninth-order polynomial, giving negligible damages below 1.5° C and very high marginal damages above that level.

Figure 1: Optimal transition paths under unconstrained welfare maximisation (cost-benefit; red) and when minimising emissions reduction costs to limit warming to 1.5° C above preindustrial (1.5° C cost-effectiveness; blue). The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



century.³¹ The carbon price is accordingly higher. We estimate an initial carbon price of US\$159/tCO2eq, rising to US\$346/tCO2eq in 2050. Both the initial level and growth rate are higher, so the price paths diverge. The substantially higher carbon price has a large effect on dirty capital disinvestment, which starts at US\$541 billion/year, is US\$442 billion/year in 2030 and remains positive beyond 2050. Cumulatively, US\$11.4 trillion in dirty capital disinvestment is made by mid-century, with US\$5.8 trillion in stranding costs (i.e., 51% of the monetary value of the disinvested dirty capital stock is lost).

The 1.5°C cost-effectiveness solution has wide confidence intervals around the carbon price and dirty capital disinvestment, while the cost-benefit solution has a wide confidence

³¹Technically, the model achieves negative emissions through negative dirty capital, but this should be seen as an abstraction. The MAC of negative emissions comes from the calibration on energy systems modelling results. Negative emissions are cost-effective in many of these models.

interval around temperatures. This reflects the different ways in which the planner responds to shocks, especially temperature shocks. The cost-benefit planner is much more willing to accept uncertainty about temperature. The 1.5°C cost-effectiveness planner is determined to keep temperatures below the limit and thus responds to temperature shocks with potentially large adjustments to carbon prices, with knock-on effects on dirty capital disinvestment.

5.2 Optimal carbon price decomposition

Figure 2 decomposes the optimal carbon price from the cost-benefit problem into its various elements as set out in Section 3 (Eq. 27). The y-axis is scaled so that a value of one corresponds to 100% of the optimal price in any given year, thus the figure plots shares.

The present value of (deterministic) marginal climate damages accounts for 112% of the optimal carbon price in 2021. This is divided into a contribution of 50 percentage points from the present value of marginal damages at the current temperature (i.e., 'current marginal damage', $\Psi c^{-\eta} y \gamma S$ in Eq. 27), and 62% from the future change in marginal damages (i.e., $-V_{SS}E$). The optimal temperature path is rapidly increasing initially, because capital inertia leads to high emissions at the start, but temperature growth slows down (Figure 1b). This explains both the large share of the future change in marginal damages initially, and the growing share of current marginal damage over time.

The disaster risk premium contributes 24% of the optimal carbon price in 2021 and this comes from the effect of temperature on disasters (we discuss this further in Section 6). The share remains relatively constant over time. Thus, avoiding climate-related macro-economic shocks is a significant motivation for reducing emissions, which persists. The contributions of the other risk premiums on temperature, capital and TFP respectively are very small.

The technical change effect starts negative overall, because the negative effect of exogenous clean technological progress (-11%) outweighs the positive effect of endogenous clean technological progress (+2%), however the situation is reversed as soon as 2024, and by 2030 technical change contributes +9% to the optimal carbon price. The dirty capital contraction effect reduces the optimal carbon price by 28% initially. This effect is inflated by dirty capital disinvestment costs, which are initially large, but it diminishes over time as the dirty capital stock settles down to a much lower and more slowly decreasing level. Opposing the dirty capital contraction effect is a small positive clean capital expansion effect (2% in 2021), which is increased by adjustment costs on clean capital investment. In the long run, the clean capital expansion effect is larger because investment flows into clean capital are larger. Figure 2: Decomposition of the optimal carbon price. Y-axis is scaled in percent of the absolute price.



5.3 Disentangling the effects of capital inertia, clean technological progress, and uncertainty

The analysis just above decomposed the optimal carbon price into a set of constituents including effects attributable to capital accumulation/decumulation, clean technological progress, and uncertainty. However, we need a different kind of analysis if we want to show the *overall* effect on optimal emissions, temperatures, etc. of modelling our three key features of the transition: capital inertia, clean technological progress, and uncertainty. This is because, in general, modifying the model to omit one of these transition dynamics affects all the elements of the optimal carbon price laid out above. For example, ignoring capital inertia not only changes the clean capital expansion and dirty capital contraction effects. Rather, by affecting the optimal temperature trajectory, it also changes current marginal damage, the future increase in marginal damage, the disaster risk premium, and so on. Therefore, to analyse what difference capital inertia, clean technological progress and uncertainty make to the optimal trajectories, in this section we compare results from the full model with a series of model versions omitting each feature in turn. Again we focus on the cost-benefit problem.

Capital inertia Figure 3 compares the full model with a version that omits adjustment costs of capital investment, and dirty capital disinvestment/stranding costs. This version is made by setting the cost parameters $\chi_c = \chi_d = \theta_1 = 0$. Adjustment/disinvestment costs make it expensive to reduce emissions quickly. Without them, it is optimal to immediately disinvest from a large portion of the dirty capital stock (US\$13 trillion of disinvestment in the first year, equal to 46% of the dirty capital stock). This leads to a huge drop in emissions of 29.5 GtCO2eq in the first year, and a flatter emissions path thereafter. Dirty capital disinvestment costs (Eq. 14) are the critical factor here. Were it possible to disinvest from dirty capital without "taking a hair cut", i.e., to recover the full value of the dirty assets, the imperative of avoiding climate damages would optimally drive nearly instantaneous disinvestment from many dirty assets. The emissions paths cross later in the 2030s, but the additional cumulative emissions in the next decade and a half that capital inertia brings about are not fully compensated this century, so optimal temperature in the scenario with adjustment costs is higher throughout.³² The difference is consistently of the order of 0.1°C. The optimal carbon price in the absence of adjustment/disinvestment costs is initially US\$129/tCO2eq, rising to US\$277/tCO2eq in 2050. Thus, the initial carbon price premium from capital inertia is 3.4%, falling to 2.2% in 2050.

Clean technological progress Figure 4 compares the full model with a model version in which clean technology costs do not fall, either exogenously or endogenously. MACs are fixed at their 2020 level, i.e., $\phi_t = \phi_0$ and $\varphi_t = \varphi_0$, $\forall t$. We find that the initial steep drop in emissions is virtually unaffected. The difference between the paths in 2030 is just 0.5GtCO2eq. It is only after 2030 that the emissions paths begin to noticeably diverge, with emissions higher in the absence of clean technological progress. This reflects the fact that clean technological progress takes time to make a big difference to MACs. Optimal emissions reductions in the early years are more strongly driven by dirty capital disinvestment costs. Higher optimal emissions lead to higher temperatures, with the optimal temperature in 2100 reaching 2.1°C. The carbon price starts at US\$138/tCO2eq with fixed MACs, thus clean technological progress provides a carbon price discount of 3.7% initially, rising to 5% in 2050. Disinvestment from dirty capital is similar in the 2020s, consistent with the similar emissions paths, but in the absence of clean technological progress there is no more dirty capital disinvestment from the early 2030s, and hence cumulative dirty capital disinvestment is lower at US\$3 trillion. Without clean technological progress, it becomes less economical to convert dirty capital to clean capital, and instead emissions/temperatures are allowed to

 $^{^{32} {\}rm Steady-state}$ temperature is the same, but adjustment/disinvestment costs make it optimal to close the gap slowly.

Figure 3: Optimal transition paths with adjustment/disinvestment costs (red) and without (blue). The planner's objective is to maximise welfare. The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



rise.

Uncertainty Figure 5 compares the full model with a model version that has no uncertainty. This is achieved by setting the volatility terms to zero in the temperature, clean/dirty capital and TFP state equations, and eliminating TFP shocks.³³ Uncertainty coupled with risk aversion leads to lower optimal emissions and temperatures in expectations than when the climate problem is treated as deterministic. Without uncertainty, the optimal temperature in 2100 is 2.1°C. The effect of uncertainty on the optimal carbon price is particularly marked – it adds 1/3 to the optimal carbon price in 2021, which is US\$100/tCO2eq without uncertainty. In 2050, it adds 23%. Accordingly, dirty capital disinvestment is substantially lower without uncertainty, at only US\$224 billion/year initially and US\$1.3 trillion cumulatively.

³³The TFP drift term is adjusted to obtain the same long-term growth path.

Figure 4: Optimal transition paths with falling clean technology costs (red) and fixed clean technology costs (blue). The planner's objective is to maximise welfare. The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



5.4 Comparison with 'straw man' model

In this section, we compare the full model with a 'straw man' model that omits capital inertia, clean technological progress, *and* uncertainty. We intentionally favour the term straw man over 'standard' as we do not want to create a misleading impression of the literature – the idea behind this model variant is that it should fairly represent a basic model of optimal management of the GHG externality. Most models allow capital to be costlessly shifted from clean sectors to dirty sectors, and many are deterministic. The treatment of clean technological progress is more nuanced, however. Some models allow clean technology costs to fall, but in most cases they fall exogenously (Gillingham et al., 2008; Nikas et al., 2018). Some models have fixed technology costs, but few if any would fix the cost parameters on 2020 data, which is what we did above in order to make the effect of falling costs clear. Therefore, for the straw man model we fix MACs at their estimated 2030 level, rather than

Figure 5: Optimal transition paths with uncertainty (red) and without uncertainty (blue). The planner's objective is to maximise welfare. The optimal path under uncertainty depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



2020. This mimics the famous McKinsey MAC curves, for example (Enkvist et al., 2007).

Figure 6 presents the comparison, with the full model in red and the straw man in blue. The straw man emissions trajectory is initially dominated by the effect of ignoring capital inertia, resulting in an immediate drop of 31GtCO2eq, which is in fact a slightly larger drop than in the variant that just omits capital inertia. Thereafter, the emissions trajectory is relatively flat, with emissions in 2100 still as high as 10GtCO2eq. This persistently high level reflects the combined effect of fixed clean technology costs and no uncertainty. The shape of the straw man emissions path translates into a more linear warming path, so temperature is initially lower than the full model optimum (due to the rapid conversion of dirty capital into clean capital), but it crosses over in the early 2050s and by 2100 the straw man optimal temperature is 2.1°C above pre-industrial. Again, this reflects the combined effect of fixed clean technology costs and no uncertainty in a higher steady-state temperature, whereas capital inertia has no effect on the steady state. In the straw man

Figure 6: Optimal transition paths for full model (red) and in straw man model without capital inertia, falling clean technology costs, or uncertainty (blue). The planner's objective is to maximise welfare. The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



model, the optimal carbon price starts at US\$100/tCO2eq. Thus, the carbon price premium in the full model relative to the straw man model is US\$18/tCO2eq or 1/3 initially, falling to 18% in 2050. There is no dirty capital disinvestment in the straw man model beyond the initial, huge adjustment, and no monetary stranding costs. Table 2 summarises the results of this section and the last. Appendix D decomposes the optimal carbon price in these different model variants.

Table 2: Summary of effects of omitting capital inertia, clean technological progress, and uncertainty respectively. Differences are expressed as (full model - model version) / model version. Note that MACs are fixed at their 2030 level in the straw man model, not 2020 as in the version with no learning.

| | Cumulative | Temperature | CO_2 price | Cumulative dirty |
|------------------------------|------------|---------------------------|--------------|-----------------------|
| | emissions | in 2100 | in 2021 | capital disinvestment |
| | 2020-2100 | | | 2020-2050 |
| Δ with no inertia | 17.6% | 0.08°C | 3.4% | -63.9% |
| Δ with no learning | -36.2% | $-0.33^{\circ}\mathrm{C}$ | -3.7% | 60.5% |
| Δ with no uncertainty | -36.5% | $-0.34^{\circ}\mathrm{C}$ | 33.3% | 257.5% |
| Δ with straw man | -34.4% | $-0.29^{\circ}\mathrm{C}$ | 32.6% | -64.7% |

5.5 Effect of delaying carbon pricing on dirty capital disinvestment

One of our main findings is that the optimal transition involves significant disinvestment from dirty assets, as the optimal speed of emissions reductions is beyond what can be achieved by just letting dirty capital depreciate and relying on autonomous improvements in energy efficiency.

Figure 7 plots cumulative disinvestment from dirty capital assets as a function of how long optimal climate policy is delayed. That is, all transitions analysed so far have optimally restricted GHG emissions from 2020 onwards. Capital inertia inhibits the capacity of the economy to make a large jump from dirty to clean capital, but the planner still imposes the optimal carbon price at the outset. Instead, here the internalisation of the GHG externality is delayed by increments of one year, as a representation of policy failures at the international and national levels. During the period of policy delay, we assume there is no anticipation of the introduction of carbon pricing, thus it is a policy shock.

The figure shows that total disinvestment from dirty capital is increasing in the length of the delay. The rate of increase is particularly fast in the case of limiting warming to 1.5°C at minimum abatement cost, because delaying emissions reductions just uses up the available carbon budget, given that temperature is a linear function of cumulative emissions (Eq. 8). Total disinvestment thus rises from US\$11.4 trillion if optimal policy starts in 2020 to US\$25.6 trillion if it is delayed by 10 years. Even a delay of one year to 2021 increases total disinvestment by US\$1.6 trillion. Dirty capital disinvestment increases more slowly when the policy objective is to maximise welfare without a temperature constraint, because the planner re-evaluates optimal long-run temperatures. Thus, despite not facing a temperature constraint, the planner still makes up for lost time with faster rates of emissions reduction.

Figure 7: Total cumulative (undiscounted) disinvestment from dirty capital as a function of how long the imposition of optimal emissions reductions is delayed. Unconstrained welfare maximisation in red; limiting warming to 1.5°C at minimum abatement cost in blue.



5.6 Cost-benefit versus cost effectiveness

Above, we compared two different formulations of the climate planning problem, unconstrained welfare maximisation (cost-benefit), and limiting expected warming to 1.5°C at minimum discounted abatement cost (1.5°C cost-effectiveness). Thus, in switching from the former to the latter, two things change: (i) a temperature constraint is imposed, and (ii) damages are effectively ignored. In this section, we isolate the effect of switching from a cost-benefit to a cost-effectiveness problem, i.e., we focus on the implications of (ii) for the optimal path. We do this by relaxing the temperature constraint in the cost-effectiveness problem so that it is just equal to the 2100 temperature that solves the cost-benefit problem.

Figure 8 shows that, since the cost-effectiveness solution is insensitive to the timing of damages, emissions are higher in the next few decades than they are in the cost-benefit solution. As a result, temperature increases slightly faster. In the second half of the century, the ordering of the emissions paths is reversed, as the cost-effectiveness solution must compensate for the initially higher emissions, which eat up the available carbon budget faster.

These differences are not large, however the differences in the optimal carbon price and dirty capital disinvestment paths are more so. The cost-effectiveness carbon price starts at US\$116/tCO2eq, which is 13% lower than the cost-benefit carbon price. It grows more quickly, so the difference between the price paths narrows to just 2.1% in 2050. The lower

Figure 8: Cost-benefit (red) and cost-effectiveness (blue) paths to achieving a common temperature in 2100. The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



carbon price results in 17% less dirty capital disinvestment initially, and 33% less cumulatively.

Comparing these results with Figure 1, the implication is that the substantially higher initial carbon price and dirty capital disinvestment required to limit warming to 1.5° C at minimum abatement cost is entirely due to the imposition of a very strict temperature constraint. Conversely, casting the policy problem as one of limiting temperature at lowest abatement cost in itself requires a lower initial carbon price and less dirty capital disinvestment (the latter in expectations, at least). Without needing to worry about damages, the planner back-loads effort until later in the century.

6 Discussion

In this section, we provide further context to and discussion of four key issues/results: (i) the level and growth rate of the optimal carbon price; (ii) clean technological progress; (iii) the risk premium on the optimal carbon price; (iv) stranded assets.

Carbon prices We estimate an optimal carbon price of US\$133/tCO2eq in 2021 to solve the cost-benefit problem. This is consistent with a recent trend towards higher optimal carbon prices in the literature. There are various reasons for this, especially lower discounting and higher climate damages, but also updating IAMs' climate dynamics (Dietz et al., 2021a). The recent study by Hänsel et al. (2020) is a good illustration of the trend. They implement various changes to DICE (the 2016 version; Nordhaus, 2017), which has an optimal carbon price in 2020 of US\$37/tCO2 (in 2010 US dollars). Reducing the pure rate of time preference from 1.5% to 0.5% and reducing the elasticity of marginal utility from 1.45 to 1 raises the optimal carbon price to US\$199/tCO2 in 2020. Our discount rate parameters are in between (1.1% and 1.35 respectively), as is our optimal carbon price. Retaining Nordhaus' preferred discount rate parameters but *inter alia* updating the DICE model's climate dynamics and increasing assumed damages, Hänsel et al. (2020) estimate a 2020 optimal carbon price of US\$82/tCO2. We use similar climate dynamics and a similar damage function calibration, but add adjustment costs, falling clean technology costs, and risk.

Besides initial carbon prices, the carbon price growth rate is also an important variable. Our optimal carbon price grows at an average rate of 3% between 2021 and 2030, falling thereafter and averaging 2.5% over the whole period 2021-2100. How this relates to GDP growth has been of particular interest ever since Golosov et al. (2014) showed that, under restrictive assumptions, carbon price and GDP growth are the same. Since GDP growth in our model is c. 2%, our carbon price grows faster. Dietz and Venmans (2019) showed in a simple analytical model that the carbon price grows faster than GDP if marginal damages are increasing in cumulative emissions, but temperature is linear in cumulative emissions. The former assumption is standard and the latter is a foundational result from climate science (IPCC, 2021). Our analysis above shows that capital inertia further contributes to the wedge between carbon price and GDP growth in the near term.

The optimal carbon price that solves the 1.5°C cost-effectiveness problem is higher at US\$159/tCO2eq in 2021, since the 1.5°C temperature constraint binds, especially under risk. It tends to grow a bit more slowly, at an average rate of 2.4% between 2021 and 2100, although in the first decade the growth rate is 3.4%. The 1.5°C cost-effectiveness problem is evaluated by energy systems models participating in the IPCC and NGFS inter-comparison

exercises. We fit our abatement cost parameters on these model results, so clearly they cannot be used to validate our carbon price estimates. However, we fit on the relationship between emissions and carbon prices, which leaves room for a different relationship between prices and time in the IPCC/NGFS models. Indeed, what is interesting about these energy systems models is that they yield lower initial carbon prices, but a much steeper trajectory, resulting in much higher prices by mid-century and beyond (Gollier, 2021). This is accompanied by higher initial emissions, and lower emissions later on, often significantly negative. Two likely reasons for this difference are that: (a) the IPCC/NGFS models tend to be solved to meet a temperature constraint in 2100 and allow overshoot in the meantime, whereas we impose the temperature constraint throughout, and (b) energy systems models tend to use higher discount rates calibrated on the market opportunity cost of capital, which leads to postponing emissions reductions.

Clean technological progress In our model, the cost of clean technology falls along the transition. Part of this process is exogenous and just depends on time. Part of it is endogenous, depending on cumulative past abatement. This dual structure is intended to capture in reduced form the diverse sources of falling clean technology costs, which include knowledge spillovers from other sectors, economies of scale and learning-by-doing in the clean technology sector, induced R&D into clean technology, and network effects. Some of these are associated with market imperfections and can be addressed by a wide range of policies including R&D subsidies, both economy-wide and specific to clean technology. The literature on clean technological progress has differentiated between R&D models and learning-bydoing models (Goulder and Mathai, 2000; Gerlagh et al., 2009). The latter models assume technologies need to be used for their costs to fall, the former need not. A pure R&D model would map on to exogenous clean technological progress in our model, with any existing policies to stimulate R&D also exogenous. A pure learning-by-doing model would map on to endogenous clean technological progress in our model. However, the dichotomy is somewhat misleading given the diverse sources of technological progress as mentioned above. Economies of scale also create a linkage between past abatement and current abatement costs, without necessarily involving an externality.

The carbon price we report corresponds to the optimal marginal abatement cost. Along the transition, it is optimal to price carbon because of (i) climate damages, which are an external cost of carbon emissions, and (ii) future abatement cost reductions, which result from current abatement. Insofar as the second component results from a different market imperfection, the first best would be to address it with a separate policy instrument such as an innovation/deployment subsidy, rather than being priced into a carbon tax (or indirectly
into the emissions quota in a system of tradeable emissions permits). In this case, our optimal carbon price would be second best, specifically it would be larger than the explicit carbon price that one would want to levy in the real world. The results of our decomposition of the optimal carbon price can be used to identify the difference. Figure 2 shows that the endogenous technical change effect is about 2% initially and reaches a maximum of about 9% in the early 2030s.

Risk premium Our results show that economic uncertainties, i.e., shocks on clean capital, dirty capital and TFP, have a very small effect on the optimal carbon price, and even macroeconomic risk has a modest effect. To intuit this result, it is helpful to discuss the climate beta (Dietz et al., 2018), i.e., the elasticity of marginal climate damages with respect to consumption. The climate beta indicates whether high marginal damages – high marginal benefits of abatement – tend to accrue in future states with high (low) consumption, in which case abatement increases (decreases) consumption risk. Our climate beta is close to one throughout the 21st century (Appendix E). It starts at just above one and decreases, non-monotonically, to about 0.85 in 2100. There are three drivers of this result. First, we assume damages are proportional to production. It is well known that in such models with 'multiplicative' damages, the climate beta will be attracted to one, as shocks on clean capital and TFP will affect consumption and marginal damages proportionally (abstracting from feedbacks via the savings rate). Second, dirty capital shocks create a positive correlation between temperature and consumption, increasing beta. Third, temperature shocks create a negative correlation between temperature and consumption, decreasing beta. The third effect dominates the second when temperature is high and the dirty capital stock is low.

In our model, intertemporal inequality aversion ($\eta = 1.35$) is hence close to the climate beta. In a model where $\eta = \beta$, a negative shock to GDP will increase marginal utility and decrease damages in equal proportion (Dietz et al., 2018; van den Bremer and van der Ploeg, 2021). As a result, the carbon price will not be affected by growth/macroeconomic uncertainty, as is the case in Golosov et al. (2014). Some authors have set $\eta < 1$ in line with the observation in capital markets that the asset/dividend ratio decreases with uncertainty (Olijslagers et al., 2021; Cai and Lontzek, 2019; Bansal et al., 2017). Others use $\eta > 1$ (Lemoine and Traeger, 2014; Cai and Lontzek, 2019; Nordhaus, 2017; van den Bremer and van der Ploeg, 2021). This is more in line with the literature on the social discount rate, and gives the more intuitive result that economic risk increases the carbon price.

Unlike economic uncertainty, the effect of climate disasters is large, increasing the optimal carbon price by 24% in 2021. However, this is mainly due to expected damage from disasters: the risk premium on the mean-preserving volatility of the damages is very small. The

effect of the Brownian motion on temperature is similarly very small. Appendix E analyses the sensitivity of our model to risk aversion. Our default assumption is that RRA = 4. In the Appendix, we test alternative values of 6 and 1.35 (the latter gives the expected utility model). The optimal carbon price varies little. Other studies investigating climate uncertainties and tipping points find similar results (Cai and Lontzek, 2019; Hambel et al., 2021; Crost and Traeger, 2014; van den Bremer and van der Ploeg, 2021).³⁴ One explanation is that consumption volatility caused by climate damages is modest compared to stockmarket volatility. van den Bremer and van der Ploeg (2021) show that the social cost of carbon more than doubles when RRA is increased from 4.32 to 6, if consumption volatility is calibrated on asset volatility. Conversely, they find almost no effect when consumption volatility is calibrated on GDP volatility. Another explanation is that our model, in keeping with the literature using recursive IAMs, assumes an optimal and immediate policy reaction to new information regarding damages and climate. A shock on temperature is followed by a rapid emissions reduction. Although adjustment costs limit how rapid it can be, the political decision comes without delay. In reality, policy comes with a delay at best and the absence of climate policy can lead to much higher risk premiums. Future research could usefully explore models of policy reaction in between a closed-loop solution with immediate policy adaptation and an open-loop solution that never adjusts the policy.

Stranded assets Our model finds it is optimal to reduce the dirty capital stock faster than its natural rate of depreciation. This can be interpreted as repurposing carbon-intensive capital for low-carbon uses. We find optimal cumulative dirty capital disinvestment of US\$4.8 trillion in the cost-benefit case and US\$11.4 trillion in the 1.5°C cost-effectiveness case. These values represent 17-41% of the total dirty capital stock in the initial period ($K_{d,0} =$ US\$28 trillion), respectively. The associated cumulative monetary stranding costs are equal to US\$1.6 trillion in the cost-benefit case and US\$5.8 trillion in the 1.5°C cost-effectiveness case. This means that 33-51% of the repurposed capital stock gets lost, respectively. As Figure 8 shows, delaying the introduction of carbon pricing would increase optimal disinvestment (and associated costs), involving almost the entire stock of dirty capital if carbon pricing is postponed to the 2030s and its imposition is not anticipated.

These results are broadly consistent with literature based on technology-rich energy models and asset-level data (see for instance Pfeiffer et al., 2018; Fofrich et al., 2020; Tong et al., 2019). These studies, often focused on specific sectors (e.g., coal- and gas-fueled electricity generation), consistently find that imposing a 1.5-2°C temperature constraint makes it

³⁴Cai and Lontzek (2019) find that increasing the *RRA* from 2 to 5 increases the SCC by a mere 3% (Fig. 7 therein). In Hambel et al. (2021), the effect is 3% for the Dell et al. (2012) damages calibration and $\eta = 1$ (Table 6 therein). The effect is barely visible in Crost and Traeger (2014), SM Fig 5.

optimal to strand a proportion of existing and planned carbon-intensive physical capital. Comparing their numerical results to ours is not trivial. Most of this applied literature provides stranding results in physical rather than monetary values (e.g., gigawatts of stranded power capacity). In addition, their sector-specific results cannot be easily translated into economy-wide stranding estimates (see Cahen-Fourot et al., 2021, for a discussion of stranding network effects). However, our results appear to be roughly in line with the results from Johnson et al. (2015) and on the lower end of the results from Edwards et al. (2022).³⁵

In addition to physical capital, reserves of fossil fuels could also be left stranded in lowcarbon scenarios. Estimates of fossil fuel reserves/resources at risk of stranding vary quite widely, from \$3 trillion to more than \$180 trillion (see Hansen, 2022, for a review), but the fact of 'unburnable carbon', i.e., the disparity between, on the one hand, the carbon embodied in available fossil fuels and, on the other hand, the 1.5-2°C cumulative carbon emissions budget, is well documented (see for instance Welsby et al., 2021). This is beyond the scope of our model, which does not include fossil fuel reserves. Given the misalignment between available fossil fuels and allowable cumulative emissions, the scarcity rent on fossil fuels in our model would be zero and would not impact on the optimal transition.

7 Conclusion

In this paper, we have studied the optimal transition to a low-carbon economy, recognising that the transition is not going to be seamless and smooth. It must be prepared for adjustment costs to the reallocation of capital from dirty to clean sectors, and shocks to the economy and climate. It also needs to account for clean technological progress, part of which is exogenous and incentivises waiting, but part of which is endogenous and incentivises acting early.

We find the optimal transition is fast, requiring a high carbon price and active stranding of dirty assets, especially if the imposition of a carbon price is delayed and unanticipated. Adjustment costs create dirty capital inertia, which in turn creates temperature inertia. Clean technological progress allows for lower optimal carbon prices, compared to a counterfactual of no progress beyond today's technology, but conversely endogenous technological progress justifies a carbon price premium that may be rather small, especially if other policy instruments are available. Uncertainty leads to lower optimal emissions and temperatures, and higher optimal carbon prices and stranded assets. However, the driver of this is expected

³⁵Johnson et al. (2015), using the MESSAGE-MACRO model, find optimal stranding of coal plants in a 2°C scenario to be between \$200 billion and \$600 billion, depending on how rapidly policies are implemented. Edwards et al. (2022), using the GCAM model, find stranded physical capital in the coal sector in the range of \$1-1.4 trillion.

damages from temperature-related macroeconomic disasters, rather than risk per se. Putting these effects together, we estimate a net premium of one third on the optimal carbon price today relative to a 'straw man' model with perfect capital mobility, fixed abatement costs and no uncertainty. Another reason to price carbon at a high level today is to avoid climate damages in the near to medium term, a basic point that is ignored if the planning problem is construed as one of meeting a temperature constraint at minimum discounted abatement cost.

References

- Acemoglu, D., Aghion, P., Bursztyn, L., Hemous, D., 2012. The Environment and Directed Technical Change. American Economic Review 102, 131–166. doi:10.1257/aer.102.1.131.
- Baldwin, E., Cai, Y., Kuralbayeva, K., 2020. To build or not to build? Capital stocks and climate policy. Journal of Environmental Economics and Management 100, 102235. doi:10.1016/j. jeem.2019.05.001.
- Bansal, R., Ochoa, M., Kiku, D., 2017. Climate change and growth risks. Technical Report. National Bureau of Economic Research.
- Barnett, M.D., 2019. A run on oil: climate policy, stranded assets, and asset prices. Ph.D. thesis. The University of Chicago.
- Barro, R.J., 2009. Rare disasters, asset prices, and welfare costs. American Economic Review 99, 243–64.
- Bertram, C., Hilaire, J., Kriegler, E., et al., 2021. NGFS Climate Scenarios Database. Technical Documentation V2.2. Technical Report. Network for Greening the Financial System. URL: https://www.ngfs.net/sites/default/files/ngfs_climate_scenario_ technical_documentation_final.pdf.
- Boston Consulting Group, 1970. Perspectives on Experience. Boston Consulting Group.
- van den Bremer, T.S., van der Ploeg, F., 2021. The risk-adjusted carbon price. American Economic Review 111, 2782–2810.
- Cahen-Fourot, L., Campiglio, E., Godin, A., Kemp-Benedict, E., Trsek, S., 2021. Capital stranding cascades: The impact of decarbonisation on productive asset utilisation. Energy Economics , 105581doi:10.1016/j.eneco.2021.105581.
- Cai, Y., 2019. Computational Methods in Environmental and Resource Economics. Annual Review of Resource Economics 11, 59–82. doi:10.1146/annurev-resource-100518-093841.
- Cai, Y., 2021. The role of uncertainty in controlling climate change, in: Oxford Research Encyclopedia of Economics and Finance. Oxford University Press.
- Cai, Y., Lontzek, T.S., 2019. The social cost of carbon with economic and climate risks. Journal of Political Economy 127, 2684–2734.
- Clarke, L., Jiang, K., Akimoto, K., et al., 2014. Assessing transformation pathways, in: Edenhofer,O., Pichs-Madruga, R., Sokona, Y., et al. (Eds.), Climate Change 2014: Mitigation of Climate

Change. Contribution of Working Group III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK and New York, NY, USA.

- Cooper, R.W., Haltiwanger, J.C., 2006. On the Nature of Capital Adjustment Costs. The Review of Economic Studies 73, 611–633. doi:10.1111/j.1467-937X.2006.00389.x. publisher: Oxford Academic.
- Coulomb, R., Lecuyer, O., Vogt-Schilb, A., 2019. Optimal Transition from Coal to Gas and Renewable Power Under Capacity Constraints and Adjustment Costs. Environmental and Resource Economics 73, 557–590. doi:10.1007/s10640-018-0274-4.
- Crost, B., Traeger, C.P., 2014. Optimal co2 mitigation under damage risk valuation. Nature Climate Change 4, 631–636.
- Cui, R.Y., Hultman, N., Edwards, M.R., et al., 2019. Quantifying operational lifetimes for coal power plants under the Paris goals. Nature Communications 10, 1–9. doi:10.1038/ s41467-019-12618-3.
- Daniel, K.D., Litterman, R.B., Wagner, G., 2019. Declining co2 price paths. Proceedings of the National Academy of Sciences 116, 20886–20891.
- David, J.M., Venkateswaran, V., 2019. The Sources of Capital Misallocation. American Economic Review 109, 2531–2567. doi:10.1257/aer.20180336.
- Dell, M., Jones, B.F., Olken, B.A., 2012. Temperature shocks and economic growth: Evidence from the last half century. American Economic Journal: Macroeconomics 4, 66–95.
- Dietz, S., Gollier, C., Kessler, L., 2018. The climate beta. Journal of Environmental Economics and Management 87, 258–274.
- Dietz, S., van der Ploeg, F., Rezai, A., Venmans, F., 2021a. Are economists getting climate dynamics right and does it matter? Journal of the Association of Environmental and Resource Economists 8, 895–921.
- Dietz, S., Rising, J., Stoerk, T., Wagner, G., 2021b. Economic impacts of tipping points in the climate system. Proceedings of the National Academy of Sciences 118.
- Dietz, S., Venmans, F., 2019. Cumulative carbon emissions and economic policy: In search of general principles. Journal of Environmental Economics and Management 96, 108–129. doi:10. 1016/j.jeem.2019.04.003.
- Drupp, M.A., Freeman, M.C., Groom, B., Nesje, F., 2018. Discounting Disentangled. American Economic Journal: Economic Policy 10, 109–134. doi:10.1257/pol.20160240.

- Duffie, D., Epstein, L.G., 1992. Asset Pricing with Stochastic Differential Utility. The Review of Financial Studies 5, 411–436. doi:10.1093/rfs/5.3.411.
- Edwards, M.R., Cui, R., Bindl, M., et al., 2022. Quantifying the regional stranded asset risks from new coal plants under 1.5°C. Environmental Research Letters 17, 024029. doi:10.1088/1748-9326/ac4ec2. publisher: IOP Publishing.
- Enkvist, P., Nauclér, T., Rosander, J., 2007. A cost curve for greenhouse gas reduction. McKinsey Quarterly 1, 34.
- Epstein, L.G., Zin, S.E., 1989. Substitution, Risk Aversion, and the Temporal Behavior of Consumption and Asset Returns: A Theoretical Framework. Econometrica 57, 937–969. doi:10.2307/1913778.
- Fofrich, R., Tong, D., Calvin, K., et al., 2020. Early retirement of power plants in climate mitigation scenarios. Environmental Research Letters 15, 094064. doi:10.1088/1748-9326/ab96d3.
- Gerlagh, R., Kverndokk, S., Rosendahl, K.E., 2009. Optimal timing of climate change policy: Interaction between carbon taxes and innovation externalities. Environmental and resource Economics 43, 369–390.
- Gerlagh, R., Liski, M., 2018. Carbon prices for the next hundred years. The Economic Journal 128, 728–757.
- Gillingham, K., Newell, R.G., Pizer, W.A., 2008. Modeling endogenous technological change for climate policy analysis. Energy Economics 30, 2734–2753.
- Gollier, C., 2021. The cost-efficiency carbon pricing puzzle.
- Golosov, M., Hassler, J., Krusell, P., Tsyvinski, A., 2014. Optimal taxes on fossil fuel in general equilibrium. Econometrica 82, 41–88.
- Gould, J.P., 1968. Adjustment Costs in the Theory of Investment of the Firm. The Review of Economic Studies 35, 47–55.
- Goulder, L.H., Mathai, K., 2000. Optimal co2 abatement in the presence of induced technological change. Journal of Environmental Economics and Management 39, 1–38.
- Hallegatte, S., 2009. Strategies to adapt to an uncertain climate change. Global Environmental Change 19, 240–247. doi:10.1016/j.gloenvcha.2008.12.003.
- Hambel, C., Kraft, H., Schwartz, E., 2021. Optimal carbon abatement in a stochastic equilibrium model with climate change. European Economic Review 132, 103642.

- Hambel, C., Kraft, H., van der Ploeg, F., 2020. Asset Pricing and Decarbonization: Diversification versus Climate Action. Discussion Paper Series 901. Department of Economics, University of Oxford.
- Hänsel, M.C., Drupp, M.A., Johansson, D.J., Nesje, F., Azar, C., Freeman, M.C., Groom, B., Sterner, T., 2020. Climate economics support for the UN climate targets. Nature Climate Change 10, 781–789.
- Hansen, T.A., 2022. Stranded assets and reduced profits: Analyzing the economic underpinnings of the fossil fuel industry's resistance to climate stabilization. Renewable and Sustainable Energy Reviews 158, 112144. URL: https://www.sciencedirect.com/science/article/pii/ S1364032122000727, doi:10.1016/j.rser.2022.112144.
- Heal, G., Millner, A., 2014. Reflections: Uncertainty and decision making in climate change economics. Review of Environmental Economics and Policy 8, 120–137.
- Heutel, G., Moreno-Cruz, J., Ricke, K., 2016. Climate engineering economics. Annual Review of Resource Economics 8, 99–118.
- Howard, P., Sylvan, D., 2021. Gauging Economic Consensus on Climate Change. Technical Report. Institute for Policy Integrity. New York University School of Law.
- Howard, P.H., Sterner, T., 2017. Few and Not So Far Between: A Meta-analysis of Climate Damage Estimates. Environmental and Resource Economics 68, 197–225. doi:10.1007/ s10640-017-0166-z.
- IEA, 2002. Beyond Kyoto. Energy Dynamics and Climate Stabilisation. Technical Report. International Energy Agency. Paris.
- IEA, 2019. World Energy Outlook 2019. Technical Report. International Energy Agency. Paris.
- IPCC, 2018. Global Warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. IPCC.
- IPCC, 2021. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK and New York, NY, USA.
- IPCC, 2022. Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge, UK and New York, NY, USA.

- Johnson, N., Krey, V., McCollum, D.L., Rao, S., Riahi, K., Rogelj, J., 2015. Stranded on a low-carbon planet: implications of climate policy for the phase-out of coal-based power plants. Technological Forecasting and Social Change 90, 89–102. doi:https://doi.org/10.1016/j. techfore.2014.02.028.
- Karydas, C., Xepapadeas, A., 2022. Climate change financial risks: Implications for asset pricing and interest rates. Journal of Financial Stability 63, 101061. doi:10.1016/j.jfs.2022.101061.
- Kemp-Benedict, E., 2018. Investing in a Green Transition. Ecological Economics 153, 218–236. doi:10.1016/j.ecolecon.2018.07.012.
- Kreps, D.M., Porteus, E.L., 1978. Temporal resolution of uncertainty and dynamic choice theory. Econometrica 46, 185–200.
- Lemoine, D., Rudik, I., 2017a. Managing climate change under uncertainty: Recursive integrated assessment at an inflection point. Annual Review of Resource Economics 9, 117–142.
- Lemoine, D., Rudik, I., 2017b. Steering the Climate System: Using Inertia to Lower the Cost of Policy. American Economic Review 107, 2947–2957. doi:10.1257/aer.20150986.
- Lemoine, D., Traeger, C., 2014. Watch your step: Optimal policy in a tipping climate. American Economic Journal: Economic Policy 6, 137–66.
- Lucas Jr, R.E., 1967. Adjustment costs and the theory of supply. Journal of political economy 75, 321–334.
- Mattauch, L., Matthews, H.D., Millar, R., Rezai, A., Solomon, S., Venmans, F., 2020. Steering the Climate System: Using Inertia to Lower the Cost of Policy: Comment. American Economic Review 110, 1231–1237. doi:10.1257/aer.20190089.
- McGlade, C., Ekins, P., 2015. The geographical distribution of fossil fuels unused when limiting global warming to 2C. Nature 517, 187–190.
- Miranda, M.J., Fackler, P.L., 2002. Applied Computational Economics and Finance. MIT Press, Cambridge, MA, USA.
- Nikas, A., Doukas, H., Papandreou, A., 2018. A detailed overview and consistent classification of climate-economy models. Understanding Risks and Uncertainties in Energy and Climate Policy: Multidisciplinary Methods and Tools for a Low Carbon Society, 1.
- Nordhaus, W.D., 1991. To slow or not to slow: The economics of the greenhouse effect. Economic Journal 101, 920–937.
- Nordhaus, W.D., 2008. A Question of Balance. Yale University Press.

Nordhaus, W.D., 2013. The Climate Casino. Yale University Press.

- Nordhaus, W.D., 2017. Revisiting the social cost of carbon. Proceedings of the National Academy of Sciences 114, 1518–1523.
- Nordhaus, W.D., Boyer, J., 2000. Warming the World: Economic Models of Global Warming. MIT press.
- Olijslagers, S., der Ploeg, R.v., van Wijnbergen, S., 2021. On current and future carbon prices in a risky world. Discussion Paper 2021-045/VI. Tinbergen Institute.
- Pfeiffer, A., Hepburn, C., Vogt-Schilb, A., Caldecott, B., 2018. Committed emissions from existing and planned power plants and asset stranding required to meet the Paris Agreement. Environmental Research Letters 13, 054019.
- van der Ploeg, F., 2018. The safe carbon budget. Climatic change 147, 47–59.
- Popp, D., 2019. Environmental Policy and Innovation: A Decade of Research. Working Paper 25631. National Bureau of Economic Research. URL: https://www.nber.org/papers/w25631, doi:10.3386/w25631.
- Rozenberg, J., Vogt-Schilb, A., Hallegatte, S., 2020. Instrument choice and stranded assets in the transition to clean capital. Journal of Environmental Economics and Management 100, 102183. doi:10.1016/j.jeem.2018.10.005.
- Semieniuk, G., Campiglio, E., Mercure, J.F., Volz, U., Edwards, N.R., 2021. Low-carbon transition risks for finance. WIREs Climate Change 12, e678. doi:10.1002/wcc.678.
- Stadler, K., Wood, R., Bulavskaya, T., et al., 2018. EXIOBASE 3: Developing a Time Series of Detailed Environmentally Extended Multi-Regional Input-Output Tables. Journal of Industrial Ecology 22, 502–515. doi:10.1111/jiec.12715.
- Taalbi, J., 2020. Evolution and structure of technological systems An innovation output network. Research Policy 49, 104010. doi:10.1016/j.respol.2020.104010.
- Timmer, M.P., Dietzenbacher, E., Los, B., Stehrer, R., de Vries, G.J., 2015. An Illustrated User Guide to the World Input–Output Database: The Case of Global Automotive Production. Review of International Economics 23, 575–605. doi:10.1111/roie.12178.
- Tol, R.S., 2011. The social cost of carbon. Annual Review of Resource Economics, Vol 2, 2010 3, 419–443.
- Tong, D., Zhang, Q., Zheng, Y., Caldeira, K., Shearer, C., Hong, C., Qin, Y., Davis, S.J., 2019. Committed emissions from existing energy infrastructure jeopardize 1.5°C climate target. Nature 572, 373–377. doi:https://doi.org/10.1038/s41586-019-1364-3.

- Traeger, C.P., 2014. A 4-stated DICE: Quantitatively addressing uncertainty effects in climate change. Environmental and Resource Economics 59, 1–37.
- UN, 2019. World Population Prospects 2019: Highlights. Technical Report ST/ESA/SER.A/423. United Nations, Department of Economic and Social Affairs, Population Division.
- UNEP, 2021. Emissions Gap Report 2021. Technical Report. United Nations Environment Programme.
- UNFCCC, 2016. Report of the Conference of the Parties on Its Twenty-First Session, Held in Paris from 30 November to 13 December 2015. Addendum. Part Two: Action Taken by the Conference of the Parties at Its Twenty-First Session. Technical Report FCCC/CP/2015/10/Add.1. United Nations Framework Convention on Climate Change. Paris.
- UNFCCC, 2021. Glasgow Climate Pact. Technical Report. United Nations Framework Convention on Climate Change.
- van der Ploeg, F., Rezai, A., 2020a. The risk of policy tipping and stranded carbon assets. Journal of Environmental Economics and Management 100, 102258. doi:10.1016/j.jeem.2019.102258.
- van der Ploeg, F., Rezai, A., 2020b. Stranded assets in the transition to a carbon-free economy. Annual Review of Resource Economics 12, 281–298.
- Vogt-Schilb, A., Meunier, G., Hallegatte, S., 2018. When starting with the most expensive option makes sense: Optimal timing, cost and sectoral allocation of abatement investment. Journal of Environmental Economics and Management 88, 210–233. doi:10.1016/j.jeem.2017.12.001.
- Weil, P., 1990. Nonexpected utility in macroeconomics. The Quarterly Journal of Economics 105, 29–42.
- Weiss, M., Junginger, M., Patel, M.K., Blok, K., 2010. A review of experience curve analyses for energy demand technologies. Technological forecasting and social change 77, 411–428.
- Welsby, D., Price, J., Pye, S., Ekins, P., 2021. Unextractable fossil fuels in a 1.5°C world. Nature 597, 230-234. URL: https://www.nature.com/articles/s41586-021-03821-8, doi:10.1038/s41586-021-03821-8.
- World Bank, 2021. State and Trends of Carbon Pricing 2021. Technical Report. The World Bank. Washington, DC.
- World Bank, 2022. World Development Indicators. Technical Report. The World Bank. Washington, DC.

Appendix A HJB equation, Itô's lemma and optimality conditions

Instantaneous utility in our model is

$$L\frac{\left(\frac{C}{L}\right)^{1-\eta}}{1-\eta},\tag{32}$$

where L is total population, growing at constant rate g_L and C is total consumption. Substituting in consumption per unit of effective labour $c = \frac{C}{L_0 e^{(g+g_L)t}}$, we obtain

$$e^{(g_L - (\eta - 1)g)t} L_0 \frac{c^{1 - \eta}}{1 - \eta}.$$
(33)

The exponent of the first factor is added to the discount rate. This transformation has two advantages. First, consumption per unit of effective labour has a steady state, and second, the transformed, larger discount rate will make the HJB equation easier to estimate. Consumption per unit of effective labour is a function of five state variables $\{k_d, k_c, S, \tilde{A}, t\}$ and two decision variables $\{i_d, i_c\}$,

$$c\left(k_{d}, k_{c}, S, \tilde{A}, t, i_{d}, i_{c}\right) = \tilde{A}\left(k_{d} + k_{c}\right)^{\alpha} e^{-\frac{\varphi(t, S)}{2}\mu(t, k_{d})^{2} - \frac{\gamma}{2}\zeta^{2}S^{2}} - i_{d} - i_{c} - \kappa\left(r\right),$$
(34)

with abatement $\mu(t, k_d) = \frac{\phi_0}{\varphi_0} - \psi_0 L_0 (1 + \tau(t)) k_d$. This justifies the value function of Equation (18) which depends on the five above mentioned states and the associated aggregator function in Equation (19).

The HJB equation is

$$max_{i_d,i_c}\left\{f + \frac{1}{dt}\mathbb{E}[dV(k_d,k_c,S,\tilde{A},t)]\right\} = 0.$$

Applying Itô's Lemma gives

$$0 = max_{i_d,i_c} \{ f \\ + V_{k_d} \left(i_d - \chi i_d^2 - (\delta + g_L + g) k_d \right) \\ + V_{k_c} \left(i_c - \chi i_c^2 - (\delta + g_L + g) k_c \right) \\ + \frac{1}{2} V_{k_d k_d} k_d^2 \sigma_d^2 + \frac{1}{2} V_{k_c k_c} k_c^2 \sigma_c^2 + V_{k_c k_d} k_d k_c \rho \sigma_c \sigma_d \\ + V_S E + \frac{1}{2} V_{SS} S^2 \sigma_S^2 \\ + V_A \left(A_0 - \tilde{A} \right) g_A + 0.5 V_{AA} \tilde{A}^2 \sigma_A^2 \\ + \lambda_A \left[V \left(\tilde{A} - \Delta \tilde{A} \right) - V \left(\tilde{A} \right) \right] \\ + V_t \}$$

with $\Delta \tilde{A} = \xi_1 (1 + \xi_2 \zeta S) \tilde{A}$ and $V (\tilde{A} - \Delta \tilde{A})$ shorthand notation for

$$V\left(k_d, k_c, S, \left(\tilde{A} - \Delta \tilde{A}\right), t\right).$$
(35)

We now turn to the explicit formula for the shadow price of capital. The first order condition for the maximum of the HJB equation with respect to dirty capital gives

$$f_c c_{i_d} + V_{k_d} \left(1 - 2\chi i_d \right) = 0.$$
(36)

The derivative of the aggregator function with respect to consumption is

$$f_c = \hat{\rho} c^{-\eta} \left[\left(1 - RRA \right) V \right]^{\frac{\eta - RRA}{1 - RRA}}.$$
(37)

From Equation (34) we have

$$c_{i_d} = -1 + \theta_1 \theta_2 r^{\theta_2 - 1}. \tag{38}$$

Combining Equations (36), (37) and (38) results in Equation (24) in the main text. The FOC for clean capital is the same, except that there will be no stranding/repurposing costs.

Next, we derive the formula for the carbon price. The shadow price of cumulative emissions (expressed in utils) is V_S . It is the decrease in welfare for an optimal emissions trajectory starting with an extra unit of cumulative emissions. Similarly, the shadow price of a unit of capital (in dollars) expressed in utils is V_{K_c} . To convert to units of effective labour, we can write $V_{K_c} = V_{k_c} \frac{\partial k_c}{\partial K_c} = \frac{V_{k_c}}{L_0 e^{(g+g_L)t}}$. Hence the carbon price, expressed in units of capital (dollars) is $p = \frac{-V_S}{V_{k_c}} L_0 e^{(g+g_L)t}$. Applying the envelope theorem with respect to S to the HJB equation results in our formula for the carbon price in Equation (27). Note that the

derivative of the aggregator function is tedious but straightforward:

$$f_{S} = \hat{\rho} c^{-\eta} c_{S} \left((1 - RRA) V \right)^{\frac{\eta - RRA}{1 - RRA}} + \hat{\rho} V_{S} \left[\frac{c^{1-\eta}}{1-\eta} \left(\eta - RRA \right) \left((1 - RRA) V \right)^{-\frac{1-\eta}{1 - RRA}} - \frac{1 - RRA}{1-\eta} \right].$$
(39)

Appendix B Parameters and initial values

| Parameter | Symbol | Value | Source | | |
|--|----------------------|-------------------------|--|--|--|
| Preferences | | | | | |
| Elasticity of marginal utility of consumption | η | 1.35 | Drupp et al. (2018) | | |
| Relative risk aversion | RRA | 4 | Barro (2009) | | |
| Utility discount rate | ρ | 0.011 | Drupp et al. (2018) | | |
| Production | | | | | |
| Initial TFP value | A_0 | 3.44 | Calibration | | |
| TFP growth rate compensating disasters | g_{A1} | 0.0037 | Barro (2009) | | |
| Steady state stabilisation rate | g_{A2} | 0.01 | Calibration | | |
| Std. dev. of TFP Brownian motion | σ_A | 0.013 | van den Bremer and van der Ploeg (2021) | | |
| Initial population | L_0 | 7.794 | UN (2019) | | |
| Population growth rate | g_L | 0.0042 | UN (2019) | | |
| Output elasticity of capital | α | 0.3 | Standard | | |
| Initial dirty capital | $K_{d,0}$ | 28 | Calibration | | |
| Initial clean capital | $K_{c,0}$ | 320 | Calibration | | |
| Depreciation rate | δ_d, δ_c | 0.04 | Calibration | | |
| Initial carbon intensity | ψ_0 | 2 | Calibration | | |
| ψ decline rate | g_ψ | 0.01 | Calibration | | |
| Adjustment cost parameter | χ_d, χ_c | 0.1 | van der Ploeg and Reza (2020a) | | |
| Std. dev. of k_d and k_c Brownian motions | σ_d, σ_c | 0.013 | van den Bremer and van der Ploeg (2021) | | |
| Correlation between k_d and k_c Brownian motions | $ ho_k$ | 0.5 | By assumption | | |
| Size of macro disasters without climate | ξ_1 | 0.2 | Calibration | | |
| Likelihood of a macro disaster | λ_A | 0.017 | Barro (2009) | | |
| Climate and emissions | | | | | |
| Temperature increase in 2020 | T_0 | $1.2^{\circ}\mathrm{C}$ | IPCC (2021) | | |
| TCRE | ζ | 0.0006 | IPCC (2021) | | |
| Climate damage function parameter | γ | 0.0077 | Howard and Sterner (2017) | | |
| Std. dev. of T Brownian motion | σ_T | 0.03 | IPCC (2021) | | |
| Dependence of macro disasters on T | ξ_2 | 0.07 | Calibration | | |
| Abatement, disinvestment and learning | | | | | |
| ϕ initial value | ϕ_0 | 0.00252 | Calibration | | |
| ϕ decline rate | g_{ϕ} | 0.69 | Calibration | | |
| φ initial value | φ_0 | 0.00004 | Estimation | | |
| φ decline rate | g_{φ} | 0.69 | Estimation | | |
| Disinvestment cost function parameter | θ_1 | 705 | Estimation | | |
| Disinvestment cost function exponent | θ_2 | 2.1 | By assumption | | |
| Exogenous learning weight parameter | ω | 0.5 | Calibration | | |
| Cumulative abatement at t_0 | M_0 | 100 | UNEP (2021) | | |
| Elasticity of MAC to cumulative abatement | ϵ | 0.1 | Estimation | | |

Table A1: Parameters and initial values in the central scenario

Appendix C Variable transformations and equations in the CompEcon toolbox

The value function is estimated over a five-dimensional 'rectangular' grid of 1600 points or nodes. The value function is approximated by a complete Chebyshev polynomial, containing 232 basis functions.³⁶

We use a transformation of variables to obtain a smoother function and a more stable solution. Instead of time, we use 'synthetic' time $\tau = 1 - e^{-g_{\tau}t}$, a transformed variable with a steady state. The transformation gives more precision for the initial decades, and less for the centuries after 2100.

The states k_d and k_c have two computational disadvantages. First, they interact strongly in determining their contribution to the value function because the marginal rate of substitution depends largely on the level of dirty capital. These strong interaction effects make the functional form of the value function more complicated. Second, k_d interacts strongly with the time state due to the time-dependent emissions intensity of the economy. For example the node on the grid corresponding to high k_d far in the future will have emissions above BAU (although E/K_d decreases over time, E/k_d increases over time). This node will have a strong curvature, while being useless to estimate an optimal path. Therefore, we use emissions $E = k_d L_0 \psi_0 (1 + \tau)$ and total capital $k = k_d + k_c$ as transformed states in a value function $\tilde{V}(E, k, S, \tilde{A}, \tau)$.

Differentiating the new variables gives

$$dE = dk_d L_0 \psi_0 (1+\tau) + k_d L_0 \psi_0 d\tau,$$
(40)

$$dk = dk_d + dk_c. \tag{41}$$

To sum up, consumption per unit of effective labour is now

$$c(E,k,S,\tilde{A},\tau) = \tilde{A}k^{\alpha}e^{-\frac{\varphi(\tau,S)}{2}(\phi_0/\phi_0 - E)^2 - \frac{\gamma}{2}\zeta^2 S^2} - i_d - i_c - \kappa(i_d),$$
(42)

with the slope of the marginal abatement curve decreasing over time as follows:

$$\varphi(\tau, S) = \varphi_0 \left[\omega M^{-\epsilon} + (1 - \omega) \frac{1}{1 + g_{\varphi} \tau} \right], \tag{43}$$

³⁶Chebyshev polynomials are linear combinations of cosinus functions with different frequencies, which can also be written as power functions. Our Chebyshev polynomial can be thought of as a polynomial containing 232 combinations of states where the sum of all powers is maximum 5 (for example $S_1^2 * S_2^2 * S_3$).

and

$$M = \frac{M_0 + \phi_0/\phi_0 \left(-g_\tau^{-1}\ln(1-\tau)\right) - (S-S_0)}{M_0},\tag{44}$$

subject to the following equations of motion:

$$\begin{split} dE &= \left[\left(i_d - \frac{\chi i_d^2}{k} \right) L_0 \psi_0 \left(1 + \tau \right) - \left(\delta + g_L + g \right) E \right] dt + E \sigma_d dW_d + \frac{E}{1 + \tau} \left(1 - \tau \right) g_\tau dt, \\ dk &= \left[i_d + i_c - \chi \frac{i_d^2 + i_c^2}{k} - \left(\delta + g_L + g \right) k \right] dt, \\ &+ \underbrace{\frac{E}{L_0 \psi_0 \left(1 + \tau \right)}}_{k_d} \sigma_d dW_d + \left(\underbrace{k - \frac{E}{L_0 \psi_0 \left(1 + \tau \right)}}_{k_c} \right) \sigma_c dW_c, \\ dS &= E dt + S \sigma_S dW_S, \\ d\tilde{A} &= \left(g_{A1} \tilde{A} + g_{A2} \left(A_0 - \tilde{A} \right) \right) dt + \sigma_A \tilde{A} dW_A - \xi_1 \left(1 + \xi_2 \zeta S \right) \tilde{A} dP, \\ d\tau &= \left(1 - \tau \right) g_\tau dt. \end{split}$$

To convert the derivatives for the carbon price decomposition we have

$$V_{Sk_d} = \tilde{V}_{Sk} \underbrace{1}_{\partial k/\partial k_d} + \tilde{V}_{SE} \underbrace{L_0 \psi_0 \left(1 - \tau\right)}_{\partial E/\partial k_d},\tag{45}$$

$$V_{Sk_c} = \tilde{V}_{Sk},\tag{46}$$

$$V_{St} = \left(\tilde{V}_{S\tau} + \tilde{V}_{SE} \underbrace{\frac{E}{1+\tau}}_{\partial E/\partial kd} \right) (1-\tau) g_{S\tau}.$$

$$(47)$$

The HJB equation is solved using a discrete time step of one year. The differential equations are approximated by $dX \cong X_t - X_{t-1}$, except for synthetic time where an exact solution is available, $\tau_{t+1} = 1 - (1 - \tau_t) e^{-g_\tau}$, and for emissions where the relationship between emissions and dirty capital is linear, $E_{t+1} = k_{d_{t+1}} L_0 \psi_0 (1 + \tau_{t+1})$.

We use the CompEcon Toolbox (Miranda and Fackler, 2002) to solve the model, and adapt the dpsolve command to include Epstein-Zin preferences. The CompEcon Toolbox applies a projection method where the value function is approximated by a polynomial, collocated on Chebyshev nodes. We adapt the toolbox to introduce complete Chebychev polynomials (Cai, 2019). Complete Chebyshev polynomials perform better than the tensorproduct Chebyshev polynomials, because higher order terms oscillate too much and lead to non-convergence of the algorithm. The polynomial is fitted to have a minimal squared deviation from the estimated value function on our 1600 grid points.

In our main specification, we use a 5-dimensional grid with 5, 4, 4, 4 and 5 Chebyshev nodes, respectively. This results in a grid with 1600 nodes. Our Chebyshev polynomial has 232 terms, which is less than the number of nodes because we leave out terms with a degree (sum of all powers) higher than 5. Figure A1 shows very similar results for a solution with 7776 nodes compared to our baseline solution with 1600 nodes.

Figure A1: Optimal transition paths comparing our baseline solution on a grid with 1600 nodes (red) with a solution using a finer grid with 7776 nodes, i.e. 6 nodes in each dimension (blue). The planner's objective is to maximise welfare. The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.



The initial guess for the value function is based on a linear-quadratic approximation of the problem. Next, we use value function iteration to fit the polynomial to the HJB equation. We do this for our model versions that converge most easily (cost-benefit, expected utility, only exogenous technical change). Finally, we start with the outcome of this model and gradually change parameters to solve more complicated models (cost-effectiveness, Epstein-Zin, endogenous technical change, etc.). These gradual parameter updates are possible with

the Newton algorithm in the Toolbox, which converges substantially faster, but requires a better initial guess.

The most difficult aspect for convergence of the algorithm is the fact that the shocks can bring the value of the state variable in the next period outside the grid, where polynomials give notoriously bad approximations. For example, with a 1.7% probability of a disaster in any given year, there is a 0.03% probability that a disaster is followed by another disaster, which brings TFP to 64% of its steady state value. We shrink the size of repeated shocks such that TFP does not decrease below this level of 64%. Similarly, we shrink the size of brownian motions at the highest node of emissions and cumulative emissions. These nodes are far from the optimal path. We also introduce bounds on both decision variables in case repeated shocks in the Monte Carlo simulation bring one of the state variables out of the grid.³⁷ These bounds, which are not binding within optimal regions of the grid also help for convergence of the HJB equation.

The Compecon Toolbox discretises the multidimensional probability density function of the Brownian Motions and jumps (our shocks on dirty and clean capital are correlated). The right hand side of the HJB equation is therefore a sum of stochastic realisations. We adapted the Compecon Toolbox to calculate each realisation in parallel using the multi-core 'parfor' loop in Matlab.

 $^{^{37}\}mathrm{We}$ adapted the CompEcon command dpsimul to that purpose.

Appendix D Optimal carbon price decomposition in different model variants

We repeated the optimal carbon price decomposition for the model versions that omit capital inertia, clean technological progress, and uncertainty, as well as the 'straw man' model. Table A2 reports differences in the share of the optimal carbon price attributable to each element, relative to the full model. A value of -10 in the table means the share of the optimal carbon price is 10 percentage points lower, for example. We compare the difference in shares in 2021 and 2050.

In the model version that omits adjustment costs, the temperature path is flatter, because emissions can instantaneously jump downwards. This intuitively results in a much smaller contribution to the optimal carbon price in 2021 from the future increase in marginal damages. Conversely, the technological change effect is much larger because the large initial abatement leads to rapid technological change. These effects are short-lived, as the 2050 comparison shows. The optimal carbon price with MACs fixed at their 2020 level naturally has a smaller technological change effect, but larger contributions from marginal damages, particularly future, given the higher temperature path. The model version with no uncertainty has no disaster risk premium by definition, while the straw man model shares features of the versions omitting adjustment costs and uncertainty, namely a much smaller contribution from the future increase in marginal damage initially, and no disaster risk premium.

| | No capital | Fixed clean | No | Straw | | |
|----------------------------------|------------|-------------|-------------|-------|--|--|
| | inertia | tech. costs | uncertainty | man | | |
| | 2021 | | | | | |
| Current marginal damage | -4.9 | 6.5 | 15.1 | 19.5 | | |
| Future increase marginal damage | -22.7 | 17.1 | 27.9 | -0.3 | | |
| Technological change effect | 19.3 | -8.5 | -3.8 | -10.2 | | |
| Clean capital expansion effect | 0.4 | 0.8 | 0.3 | 1.0 | | |
| Dirty capital contraction effect | 10.0 | -22.1 | -21.5 | 8.0 | | |
| Disaster risk premium | -1.8 | 5.8 | -16.9 | -16.9 | | |
| | 2050 | | | | | |
| Current marginal damage | -2.6 | -5.0 | 3.6 | 5.0 | | |
| Future increase marginal damage | 2.7 | 7.4 | 20.4 | 21.8 | | |
| Technological change effect | -0.6 | -5.9 | -2.2 | -5.2 | | |
| Clean capital expansion effect | -0.1 | 0.8 | -0.2 | -0.6 | | |
| Dirty capital contraction effect | -0.4 | -1.1 | -2.1 | -1.5 | | |
| Disaster risk premium | 0.6 | 1.3 | -20.1 | -20.7 | | |

Table A2: Differences in the share of the optimal carbon price attributable to each element, relative to the full model. Units are percentage points.

Appendix E The climate beta and the effect of risk aversion

Figure A2 plots the climate beta over this century, as estimated by our model. The climate beta (Dietz et al., 2018) may be calculated as

$$\beta_t = \frac{\operatorname{Cov}\left(\ln c_t, \ln D_{T_t}\right)}{\operatorname{Var}\left(\ln c_t\right)},$$

where marginal damage in our model $D_{T_t} = Y_t \zeta^2 \gamma T_t$. The climate beta is close to one throughout.





To investigate the effect of risk aversion we compare two models with relative risk aversion (RRA) of 1.35 and 6 respectively. The low value of RRA equals the intertemporal inequality aversion and corresponds therefore to expected utility. Increasing the risk averseness gives an insight in the effect of mean-preserving uncertainty. In the limit case where RRA converges to zero, uncertainty is not priced, only expected damages matter. RRA equals 4 in our main analysis. Figure A3 shows that increasing risk aversion has only a mild effect on carbon prices and emission trajectories.

Figure A3: Optimal transition paths comparing a low relative risk aversion of 1.35 (red) with a high relative risk aversion of 6 (blue). The planner's objective is to maximise welfare. The optimal path depends on the stochastic realisations along the path. The solid lines represent the mean of 5000 Monte Carlo runs. The shaded areas represent the 10-90% confidence interval.

